Registration	No.	
--------------	-----	--

SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM S-8

REGISTRATION STATEMENT
UNDER THE SECURITIES ACT OF 1933

CLEVELAND-CLIFFS INC

(Exact name of registrant as specified in its charter)

Ohio (State or other jurisdiction of incorporation or organization)

34-1464672 (I.R.S. Employer) Identification No.)

1100 Superior Avenue Cleveland, Ohio 44114-2589 (Address of principal executive offices including zip code)

NORTHSHORE MINING COMPANY AND SILVER BAY POWER COMPANY RETIREMENT SAVINGS PLAN (Full title of the plan)

JOHN E. LENHARD, ESQ.
Secretary & Assistant General Counsel
18th Floor, Diamond Building
1100 Superior Avenue
Cleveland, Ohio 44114-2589
(Name and address of agent for service)

(216) 694-5700

(Telephone number, including area code, of agent for service)

<TABLE>

CALCULATION OF REGISTRATION FEE

<pre><caption> Title of securities to be registered(1)</caption></pre>	Amount to be registered	Proposed maximum offering price per share	Proposed maximum aggregate offering price	Amount of registration fee
<s> Common Shares</s>	<c></c>	<c></c>	<c></c>	<c></c>
par value \$1.00 per share	200,000	\$35.0625(2)	\$7,012,500.00(2)	\$2,418.11

<FN>

- (1) Pursuant to Rule 416(c) under the Securities Act of 1933, this registration statement also covers an indeterminate amount of interests to be offered pursuant to the Northshore Mining Company and Silver Bay Power Company Retirement Savings Plan.
- (2) Pursuant to Rule 457(h) under the Securities Act of 1933, this estimate is made solely for the purpose of calculating the amount of the registration fee and is based on the average of the high and low prices of the Common Shares on the New York Stock Exchange on November 25, 1994.

</TABLE>

PART II

ITEM 3. INCORPORATION OF DOCUMENTS BY REFERENCE.

The following documents heretofore filed by Cleveland-Cliffs Inc (the "Company") and the Northshore Mining Company and Silver Bay Power Company Retirement Savings Plan (the "Plan") with the Securities and Exchange Commission are incorporated herein by reference:

- (1) Annual Report of the Company on Form 10-K for the fiscal year ended December 31, 1993;
- (2) Quarterly Reports of the Company on Form 10-Q for the fiscal quarters ended March 31, 1994, June 30, 1994, and September 30, 1994;
- (3) Current Report of the Company on Form 8-K, dated October 13, 1994;
- (4) The description of the Company's Common Stock, par value \$1.00 per share, contained in the Company's Registration Statement filed pursuant to Section 12 of the Securities Exchange Act of 1934 and any amendments and reports filed for the purpose of updating that description; and
- (5) Rights Agreement dated September 8, 1987, and amended and restated as of November 19, 1991, between the Company and Society National Bank (successor to Ameritrust Company National Association) (filed as Exhibit 4.2 to the Company's Current Report on Form 8-K on November 20, 1991).

All documents that shall be filed by the Company and the Plan pursuant to Sections 13(a), 13(c), 14 and 15(d) of the Securities Exchange Act of 1934 subsequent to the filing of this registration statement and prior to the filing of a post-effective amendment indicating that all securities offered under the Plan have been sold or deregistering all securities then remaining unsold thereunder shall be deemed to be incorporated herein by reference and shall be deemed to be a part hereof from the date of filing thereof.

ITEM 6. INDEMNIFICATION OF DIRECTORS AND OFFICERS.

Article IV of the Regulations of the Company (the "Regulations") provides as follows:

Section 1. INDEMNIFICATION. The Company shall indemnify, to the full extent then permitted by law, any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative, by reason of the fact that he is or was a director, officer, employee or agent of the Company, or is or was serving at the request of

2

the Company as a director, trustee, officer, employee or agent of another corporation, domestic or foreign, nonprofit or for profit, partnership, joint venture, trust or other enterprise; provided, however, that the Company shall indemnify any such agent (as opposed to any director, officer or employee) of the Company to an extent greater than that required by law only if and to the extent that the Directors may, in their discretion, so determine. The indemnification provided hereby shall not be deemed exclusive of any other rights to which those seeking indemnification may be entitled under any law, the Articles of Incorporation or any agreement, vote of shareholders or of disinterested Directors or otherwise, both as to action in official capacities

and as to action in another capacity while he is a Director, officer, employee or agent, and shall continue as to a person who has ceased to be a Director, trustee, officer, employee or agent, and shall inure to the benefit of heirs, executors and administrators of such a person.

Section 2. INSURANCE. The Company may, to the full extent then permitted by law and authorized by the Directors, purchase and maintain insurance on behalf of any persons described in Section 1 of this Article IV against any liability asserted against and incurred by any such person in any such capacity, or arising out of his status as such, whether or not the Company would have the power to indemnify such person against such liability.

The indemnification provided by the Regulations shall not be deemed exclusive of any other rights to which any person indemnified may be entitled as a matter of law or otherwise both as to action in his or her official capacity and as to action in another capacity while holding such office and shall continue as to a person who has ceased to be a director, officer, employee or agent and shall inure to the benefit of the heirs, executors and administrators of such a person.

Section 1701.13(E) of the General Corporation Law of the State of Ohio also permits indemnification of directors and officers of an Ohio corporation.

The directors and officers of the Company are covered by insurance policies issued by Continental Casualty Company and Federal Insurance Company, which insure the directors and officers of the Company against certain liabilities (excluding fines and penalties imposed by law) that might be incurred by them in such capacities and insure the Company for amounts that may be paid by the Company to indemnify its directors covered by the policies (up to the limits of such policies).

The Company has entered into indemnification agreements with its directors that would require the Company, subject to any limitations on the $\,$

3

maximum permissible indemnification that may exist at law, to indemnify a director for claims that arise from his or her capacity as a director.

ITEM 8. EXHIBITS.

- 4(a) Amended Articles of Incorporation of the Company (filed as Exhibit 3(a) to the Company's Annual Report on Form 10-K for the fiscal year ended December 31, 1990, and incorporated herein by reference)
- (b) Regulations of the Company (filed as Exhibit 3(b) to the Company's Annual Report on Form 10-K for the fiscal year ended December 31, 1990, and incorporated herein by reference)
- (c) Rights Agreement dated September 8, 1987, and amended and restated as of November 19, 1991, between the Company and Society National Bank (successor to Ameritrust Company National Association) (filed as Exhibit 4.2 to the Company's Current Report on Form 8-K on November 20, 1991, and incorporated herein by

reference)

- (d) Northshore Mining Company and Silver Bay Power Company Retirement Savings Plan, dated October 3, 1994
- (e) T. Rowe Price Trust Company Trust Agreement, dated October 10, 1994
- 23 Consent of Independent Auditors
- 24 Powers of Attorney

UNDERTAKING:

The undersigned registrant will submit the Plan and any amendments thereto to the Internal Revenue Service and will make all changes required by the Internal Revenue Service in order to qualify the Plan.

ITEM 9. UNDERTAKINGS.

- (a) The undersigned registrant hereby
 undertakes:
- which offers or sales are being made, a post-effective amendment to this registration statement: (i) to include any prospectus required by Section 10(a)(3) of the Securities Act of 1933; (ii) to reflect in the prospectus any facts or events arising after the effective date of the registration statement (or the most recent post-effective amendment thereof) which, individually or in the aggregate, represent a fundamental change in the information set forth in the registration statement; (iii) to include any material information with respect to the plan of distribution not previously disclosed in the registration statement or any material change to such information in the registration statement; PROVIDED,

4

HOWEVER, that paragraphs (a)(1)(i) and (a)(1)(ii) do not apply if the registration statement is on Form S-3 or Form S-8, and the information required to be included in a post-effective amendment by those paragraphs is contained in periodic reports filed by the registrant pursuant to Section 13 or Section 15(d) of the Securities Exchange Act of 1934 that are incorporated by reference in the registration statement.

- (2) That, for the purpose of determining any liability under the Securities Act of 1933, each such post-effective amendment shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
- (3) To remove from registration by means of a post-effective amendment any of the securities being registered which remain unsold at the termination of the offering.
- (b) The undersigned registrant hereby undertakes that, for purposes of determining any liability under the Securities Act of 1933, each filing of the registrant's annual report pursuant to Section 13(a) or Section 15(d) of the Securities Exchange Act of 1934 (and, where applicable, each filing of an employee benefit plan's annual report pursuant to Section 15(d) of the Securities Exchange Act of 1934) that is incorporated by reference in the registration statement shall be deemed to be a new registration statement relating to the securities offered therein, and the offering of such securities at that time shall be deemed to be the initial bona fide offering thereof.
 - (c) Insofar as indemnification for

liabilities arising under the Securities Act of 1933 may be permitted to directors, officers and controlling persons of the registrant pursuant to the foregoing provisions, or otherwise, the registrant has been advised that in the opinion of the Securities and Exchange Commission such indemnification is against public policy as expressed in the Act and is, therefore, unenforceable. In the event that a claim for indemnification against such liabilities (other than the payment by the registrant of expenses incurred or paid by a director, officer or controlling person of the registrant in the successful defense of any action, suit or proceeding) is asserted by such director, officer or controlling person in connection with the securities being registered, the registrant will, unless in the opinion of its counsel the matter has been settled by controlling precedent, submit to a court of appropriate jurisdiction the question whether such indemnification by it is against public policy as expressed in the Act and will be governed by the final adjudication of such issue.

5

SIGNATURES

Pursuant to the requirements of the Securities Act of 1933, the registrant certifies that it has reasonable grounds to believe that it meets all of the requirements for filing this registration statement on Form S-8 and has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Cleveland, State of Ohio, on this 30th day of November, 1994.

CLEVELAND-CLIFFS INC

By: /s/ John S. Brinzo

John S. Brinzo
Senior Executive-Finance

6

Pursuant to the requirements of the Securities Act of 1933, this registration statement has been signed by the following persons in the capacities and on the dates indicated.

<TABLE> <CAPTION>

Signatures	Title	Date
<s> *M. T. Moore M. T. Moore</s>	<c> Chairman, President, and Chief Executive Officer and Director (Principal Executive Officer)</c>	<c> November 30, 1994</c>
*J. S. Brinzo J. S. Brinzo	Senior Executive-Finance (Principal Financial Officer)	November 30, 1994
*R. Emmet R. Emmet	Vice President and Controller (Principal Accounting Officer)	November 30, 1994
*R. S. Colman	Director	November 30, 1994

R. S. Colman		
*J. D. Ireland III	Director	November 30, 1994
J. D. Ireland III		
*G. F. Joklik	Director	November 30, 1994
G. F. Joklik		
*E. B. Jones	Director	November 30, 1994
E. B. Jones		
*L. L. Kanuk	Director	November 30, 1994
L. L. Kanuk		
*S. B. Oresman	Director	November 30, 1994
S. B. Oresman		
*A. Schwartz	Director	November 30, 1994
A. Schwartz		
*S. K. Scovil	Director	November 30, 1994
S. K. Scovil		
*J. H. Wade	Director	November 30, 1994
J. H. Wade		
*A. W. Whitehouse	Director	November 30, 1994

</TABLE>

7

A. W. Whitehouse

* This registration statement has been signed on behalf of the above-named directors and officers of the Company by John E. Lenhard, Secretary of the Company, as attorney-in-fact pursuant to powers of attorney filed with the Securities and Exchange Commission as Exhibit 24 to this registration statement.

DATED: November 30, 1994 By: /s/ John E. Lenhard

John E. Lenhard, Attorney-in-Fact

8

THE PLAN. Pursuant to the requirements of the Securities Act of 1933, the Plan has duly caused this registration statement to be signed on its behalf by the undersigned, thereunto duly authorized, in the City of Cleveland, State of Ohio, on this 30th day of November, 1994.

By: /s/ Cynthia B. Bezik Cynthia B. Bezik

Treasurer

9

EXHIBIT INDEX

<TABLE> <CAPTION>

CCAPITON			Page Number
	Exhibit Number	Dubibit Description	in Sequentially
	Number	Exhibit Description	Numbered Copy
	<s></s>	<c></c>	<c></c>
	4 (a)	Amended Articles of Incorporation of the Company (filed as Exhibit 3(a) to the Company's Annual Report on Form 10-K for the fiscal year ended December 31, 1990, and incorporated herein by reference)	Not Applicable
	4 (b)	Regulations of the Company (filed as Exhibit 3(b) to the Company's Annual Report on Form 10-K for the fiscal year ended December 31, 1990, and incorporated herein by reference)	Not Applicable
	4 (c)	Rights Agreement dated September 8, 1987, and amended and restated as of November 19, 1991, between the Company and Society National Bank (successor to Ameritrust Company National Association) (filed as Exhibit 4.2 to the Company's Current Report on Form 8-K on November 20, 1991, and incorporated herein by reference)	
	4 (d)	Northshore Mining Company and Silver Bay Power Company Retirement Savings Plan, dated October 3, 1994	11
	4 (e)	T. Rowe Price Trust Company Trust Agreement, dated October 10, 1994	57
	23	Consent of independent auditors	70

 24 | Power of Attorney | 71 |10

NORTHSHORE MINING COMPANY AND SILVER BAY POWER COMPANY RETIREMENT SAVINGS PLAN

Sequential Page 11

<TABLE> <CAPTION>

NORTHSHORE MINING COMPANY AND SILVER BAY POWER COMPANY RETIREMENT SAVINGS PLAN

TABLE OF CONTENTS

<s></s>	<c></c>	ARTICLE I
		NAME AND PURPOSE OF PLAN 1
		ARTICLE II
		DEFINITIONS 1
2.1	ACCOUNT	
2.2	AFFILIATED COMPANY	
2.3		
2.4		
2.5	BOARD OF DIRECTORS	
2.6	CLIFFS STOCK	
2.7	CLIFFS STOCK FUND	
2.8	CODE	
2.9	COMPANY	
2.10	COMPANY CONTRIBUTIONS	
2.11	COMPANY MATCHING CONTRIBUTIONS	
2.12		
2.13	EFFECTIVE DATE	
2.14		
2.15	ERISA	
2.16		
2.17		
2.18		
2.19		
2.20		
2.21	PLAN ADMINISTRATOR	
2.22		
2.23		
2.24	QUALIFIED NON-ELECTIVE CONTRIBUTIONS	
2.25		
2.26	ROLLOVER CONTRIBUTIONS	
2.27	TERMINATION OF EMPLOYMENT or TERMINATES EMPLOYMENT	
2.28	TOTAL DISABILITY	
2.29	TRANSFER CONTRIBUTIONS	
2.30	TRUST AGREEMENT	4
2.31	TRUSTEE	
2.32	TRUST FUND	
2.33	VALUATION DATE	4

 | |(i)

ARTICLE III

		PARTICIPATION 4
2 1		
3.1 3.2	WHO MAY BECOME A PARTICIPANT	
3.3	DETERMINATION OF PARTICIPANTS	
3.4	CHANGES IN EMPLOYMENT STATUS: TRANSFERS OF EMPLOYMENT .	
J • 1	CHRISTO IN EM ESTERNI STRING, ITAMOLERO SI EM ESTERNI .	
	ADMICO	D. 711
	ARTICL	E IV
		CONTRIBUTIONS 5
4.1	PARTICIPANT CONTRIBUTIONS	
4.2	COMPANY MATCHING CONTRIBUTIONS	
4.3	CONTRIBUTION LIMITATIONS	
4.4	QUALIFIED NON-ELECTIVE CONTRIBUTIONS	
4.5	LIMITATION ON ANNUAL ADDITIONS	
4.6	LIMITATION ON COMBINED BENEFITS AND CONTRIBUTIONS UNDER A	LL
	DEFINED BENEFIT PLANS AND DEFINED CONTRIBUTION PLANS OF	
	COMPANY AND ANY RELATED CORPORATION	
4.7	EXCLUSIVE BENEFIT	
4.8	COMPANY'S OBLIGATIONS	
4.9	ROLLOVER CONTRIBUTIONS	
	350070	TP V
	ARTIC	TE V
	PARTI	CIPANTS' ACCOUNTS
5.1	PARTICIPANT'S SUBACCOUNTS	
5.2	ALLOCATION OF CONTRIBUTIONS	
5.3	VESTING OF PARTICIPANT'S ACCOUNT	
5.4	TRANSFER OF ACCOUNTS UPON TRANSFER OF EMPLOYMENT	
	ARTICL	E VI
	INVES	TMENT OF ACCOUNTS 17
6.1	INVESTMENT CATEGORIES	
6.2	CONTINUING INVESTMENT	
6.3	CHANGE OF INVESTMENT HOLDINGS	
6.4	INTERIM INVESTMENT	
6.5	VALUATION OF PARTICIPANT'S ACCOUNTS	

	(ii)			
	(11)			
\b\	ARTICL			
	DIGHDIDI	MION EDOM MDUGH BUND 10		
	DISTRIBU	TION FROM TRUST FUND 19		
7.1	WHEN ACCOUNT BECOMES DISTRIBUTABLE			
7.2	TIME AND FORM OF PAYMENT			
7.3				
7.4	DIRECT ROLLOVER OF DISTRIBUTIONS			
7.5	SPENDTHRIFT TRUST PROVISIONS			
7.6 7.7	DISTRIBUTION IN THE EVENT OF DEATH			
7.7	BENEFICIARIES			
7.9	PARTICIPANT LOANS			
, • 9	IIIMITOTIMI BOIMO			
	ARTICLE	VIII		
	ווחדק	CIARY OBLIGATIONS 28		
8.1	GENERAL FIDUCIARY DUTIES			
8.2 8.3	ALLOCATION OF FIDUCIARY RESPONSIBILITY			
0.3	COMPENSATION AND EXPENSES OF FIDUCIARIES			
		7. 7.7		
	ARTICL	E 1X		
	PLA	N ADMINISTRATOR 28		

9.1 9.2 9.3 9.4 9.5	INFORMATION TO BE MADE AVAILABLE TO PLAN ADMINISTRATOR	18 19 19 10
	ARTICLE X	
	TRUST FUND 3	1
10.1		1 2
	ARTICLE XI	
	CONTINUANCE, TERMINATION AND	
	·	3
11.1 11.2	MERGER, CONSOLIDATION OR TRANSFER OF ASSETS OR	3
11.3 11.4 		

 AMENDMENTS TO PLAN | 3 4 3 4 || | (iii) | |
	ARTICLE XII	(C>
	AKIICHE AII	
	MISCELLANEOUS	5
12.1	BENEFITS TO BE PROVIDED SOLELY FROM THE TRUST FUND	5
12.2		5
12.3		5
12.4 12.5	JURISDICTION	6
12.6		6
	ADMITATE WITT	
	ARTICLE XIII	
	TOP HEAVY PROVISIONS	6
13.1	DEFINITIONS	6
13.2	COMPENSATION AND LIMITATION THEREON	8
13.3		8
13.4	MINIMUM ALLOCATION	8
	ARTICLE XIV	
	SPECIAL PROVISIONS REGARDING ACQUISITION EMPLOYEES . 3	9
14.1	GENERAL	9
14.2	EMPLOYMENT	9
14.3	TRANSFER OF ASSETS AND LIABILITIES	9
14.4	CONTINUATION OF PORTION OF CYPRUS SAVINGS PLAN	0
	EXHIBIT "A"	
</TABLE>

(iv)

NORTHSHORE MINING COMPANY AND SILVER BAY POWER COMPANY RETIREMENT SAVINGS PLAN

ARTICLE I

Northshore Mining Company, by execution of this agreement, establishes a plan to be known as the Northshore Mining Company and Silver Bay Power Company Retirement Savings Plan (the "Plan"), effective as of the day following the "Closing Date" of the transactions contemplated by the Stock Purchase Agreement by and between Cleveland-Cliffs Inc and Cliffs Minnesota Minerals Company and Cyprus Amax Minerals Company, as amended (the "Stock Purchase Agreement"), as "Closing Date" is defined in such Stock Purchase Agreement; and Silver Bay Power Company, by execution of this Agreement, adopts the Plan effective as of the Closing Date. The Plan is created for the exclusive benefit of Participants and their Beneficiaries. The Plan is intended to qualify under Sections 401(a) and 401(k) of the Code as a qualified cash or deferred arrangement, and the Trust created under the Plan is intended to be exempt under Section 501(a) of the Code.

ARTICLE II

DEFINITIONS

When used in the Plan, the following words will have the following meanings, unless the context clearly indicates otherwise:

- 2.1 "Account", unless otherwise indicated, means a Participant's entire interest in the Trust Fund.
- "Affiliated Company" means any corporation that is a member of the "controlled group of corporations" of which the Company is also a member (as such term is defined in Section 1563(a) of the Code without regard to Section 1563(a) (4) of the Code).
- 2.3 "After-Tax Contributions" mean the amounts contributed by the Participant pursuant to Section 4.1[b].
- 2.4 "Beneficiary" means the person who, under this Plan, becomes entitled to receive a Participant's Account upon his death, including when applicable the surviving spouse.
- 2.5 "Board of Directors", unless otherwise specified, means the Board of of Northshore Mining Company.
- 2.6 "Cliffs Stock" means the common stock, par value \$1.00, of Cleveland-Cliffs Inc.
- 2.7 "Cliffs Stock Fund" means the fund invested solely in Cliffs Stock.
- 2.8 "Code" means the Internal Revenue Code of 1986, as amended, or any successor statute of similar purpose.
- 2.9 "Company" means Northshore Mining Company, and any Affiliated Company that adopts this Plan with the approval of the Board of Directors (which Companies shall be listed on Exhibit A to the Plan), and any successor in interest resulting from merger, consolidation or transfer of substantially all of the Company's assets that may expressly agree in writing to continue this Plan.
- 2.10 "Company Contributions" mean the amounts contributed under the Plan by the Company as provided in Article IV, including Company Matching Contributions; and shall also include the applicable portion of a Participant's Transfer Contributions.
- 2.11 "Company Matching Contributions" mean the amounts contributed by the Company pursuant to Section 4.2 and allocated pursuant to Section 5.2[a].
- "Compensation" means the basic salary paid to a Participant for services rendered to the Company during the Plan Year and includes basic earnings paid to a Participant during the Plan Year that was deferred from a previous year and any salary reduction contributions made under this Plan but does not include any basic salary for the Plan Year that is deferred to a subsequent year, overtime, bonuses, commissions, moving allowances or any other extraordinary compensation. In addition to other applicable limitations which may be set forth in the Plan and notwithstanding any other contrary provision of the Plan, Compensation taken into account under the Plan for a Plan Year shall not exceed \$150,000, as adjusted for changes in the cost-of-living as provided in Sections 401(a)(17)(B) and 415(d) of the Code. In determining the Compensation of a Participant for purposes of this limitation, the rules of Section 414(q)(6) of the Code shall apply, except that in applying such rules the term "family" shall only include the spouse of the Participant and any lineal descendants of the Participant who have not attained age 19 before the close of the Plan Year.
- 2.13 "Effective Date" means the day following the "Closing Date" of the transactions contemplated by the Stock Purchase Agreement by and between

Cleveland-Cliffs Inc and Cliffs Minnesota Minerals Company and Cyprus Amax Minerals Company, as amended (the "Stock Purchase Agreement"), as "Closing Date" is defined in such Stock Purchase Agreement.

- 2.14 "Employee" means any person now or hereafter in the employ of the Company at Company locations in the United States, and United States citizens at other locations while being paid on the United States payroll, including officers of the Company, but excluding (i) employees included in a unit of employees covered by a collective bargaining agreement between employee representatives and the Company if there is evidence that retirement benefits were the subject of good faith bargaining between such employee representatives and the Company, (ii) directors who are not employed by the Company in any other capacity, (iii) independent contractors, (iv) Leased Employees, and (v) any employee who is a self-employed individual or owner-employee within the meaning of Section 401(c) of the Code with respect to his employment with the Company.
- 2.15 "ERISA" shall mean the Employee Retirement Income Security Act of 1974, as amended.
- 2.16 "Leased Employee" means any person (other than an employee of the recipient) who pursuant to an agreement between the recipient and any other person (a "leasing organization") has performed services for the recipient (or for the recipient and related persons determined in accordance with Section $414\,(n)\,(6)$

-2-

of the Code) on a substantially full-time basis for a period of at least one year, and where such services are of a type historically performed by employees in the business field of the recipient employer. Contributions or benefits provided a Leased Employee by the leasing organization which are attributable to services performed for the recipient employer shall be treated as provided by the recipient employer.

A Leased Employee shall not be considered an employee of the recipient if: (i) such Leased Employee is covered by a money purchase pension plan providing: (A) a non-integrated employer contribution rate of at least ten percent of "compensation" (as defined in Section 415(c)(3) of the Code) but including amounts contributed by the employer pursuant to a salary reduction agreement which are excludable from the employee's gross income under Section 125, 402(a)(8), 402(h) or 403(b) of the Code, (B) immediate participation, and (C) full and immediate vesting; and (ii) Leased Employees do not constitute more than 20 percent of the recipient's "Non-Highly Compensated Employee" (as defined in Section 4.3[c][4]) work force.

- 2.17 "Normal Retirement Age" means age 65. Notwithstanding any other provision of the Plan to the contrary, an Employee shall be 100 percent vested in his Account upon attainment of age 65.
- 2.18 "Participant" means any Employee who has become a Participant under this Plan. Participation will cease upon distribution of a Participant's entire vested Account after Termination of Employment.
- 2.19 "Participant Contributions" mean the amounts contributed under the Plan by Participants as provided in Article IV, including Pre-Tax Contributions, After-Tax Contributions and Rollover Contributions; and shall also include the applicable portion of a Participant's Transfer Contributions.
- 2.20 "Plan" means the plan maintained under this document and all subsequent amendments to it.
- 2.21 "Plan Administrator" means the person or persons appointed by the Board of Directors whose duties are specified in this Plan.
- 2.22 "Plan Year" means the calendar year; provided, however, that the first Plan Year shall be a "Short Plan Year" commencing on the Effective Date and ending December 31, 1994.
- 2.23 "Pre-Tax Contributions" mean the amounts contributed by the Participant pursuant to Section 4.1[b].
- 2.24 "Qualified Non-Elective Contributions" mean the amounts contributed by the Company pursuant to Section 4.4.
- 2.25 "Related Corporation" means any corporation which is a member of a controlled group of corporations of which the Company is also a member, as determined under Section 1563(a) of the Code, without regard to Sections 1563(a) (4) and 1563(e) (3) (C) of the Code. Furthermore, the term shall include any trade or business (whether or not incorporated) which is a member of a group under common control of which the Company is also a member, as determined

under Section 414(c) of the Code. The term shall also include each organization which is a member of an affiliated service group of which the Company is also a member,

-3-

as determined under Section $414\,(m)$ of the Code. Finally, the term shall include any entity, other than the Company, which is required to be aggregated with the Company under Section $414\,(o)$ of the Code.

- 2.26 "Rollover Contributions" mean the amounts contributed by the Participant pursuant to Section 4.9.
- 2.27 "Termination of Employment" or "Terminates Employment" means a termination of the employer-employee relationship between an Employee and the Company for any reason. Transfer to employment (i) with a Related Corporation that does not maintain this Plan or (ii) to an ineligible class of employees, will not be considered a Termination of Employment, and Plan participation will be governed by the provisions of Section 3.4.
- 2.28 "Total Disability" means a disability that renders a Participant unable to perform satisfactorily the usual duties of his employment with the Company, as determined by a physician selected by the Plan Administrator, and which results in his Termination of Employment with the Company.
- 2.29 "Transfer Contributions" mean the amounts transferred to the Plan from the Cyprus Amax Minerals Company Savings Plan & Trust, as described in Article XIV.
- 2.30 "Trust Agreement" means the agreement entered into between Northshore Mining Company and the Trustee, which agreement is incorporated herein by reference.
- 2.31 "Trustee" means the person or persons appointed by the Board of Directors as the trustee of the Trust Fund and any duly appointed and qualified successor trustee.
- 2.32 "Trust Fund" means the assets of the Plan from which benefits will be paid and includes all income of any nature earned by such trust fund and all changes in fair market value.
- 2.33 "Valuation Date" means the date investments are valued and shall occur on each business day on which the Plan's recordkeeper and the New York Stock Exchange are open for business.

ARTICLE III

PARTICIPATION

3.1 WHO MAY BECOME A PARTICIPANT: Any Employee who is an Employee on the Effective Date will become a Participant on such Effective Date; provided, however, such Employee has completed the appropriate enrollment form and timely filed such form with the Plan Administrator. Any other Employee will become a Participant as of the first pay period of the calendar month next following the date the Employee commences or recommences employment with the Company; provided, however, such Employee has completed the appropriate enrollment form and timely filed such form with the Plan Administrator. Any Employee who does not become a Participant when first eligible to do so may become a Participant as of the first pay period of the calendar month next following the Employee's completion of the appropriate enrollment form and timely filing of such form with the Plan

-4-

Administrator. Any Employee who is a Participant will continue to be a Participant upon restatement or amendment of the Plan, unless the restatement or amendment specifically excludes the Employee from participation.

3.2 DETERMINATION OF PARTICIPANTS: The Plan Administrator will determine when Employees become eligible to participate in the Plan and will provide such Employees with the necessary enrollment forms to commence participation in the Plan. An Employee may enroll during the month he first becomes eligible to participate in the Plan or any month thereafter by filing the completed enrollment forms with the Plan Administrator within the required time before the month his election to participate in the Plan begins. Participation will

begin the first pay period in the month following the Employee's completion of the enrollment form and timely filing of it with the Plan Administrator.

- 3.3 PARTICIPATION UPON REEMPLOYMENT: An Employee may become a Participant upon his reemployment as an Employee or transfer to an eligible class of Employees, as described in Section 2.14, on the first pay period in the month following the Employee's completion of the appropriate enrollment form and timely filing of it with the Plan Administrator.
- 3.4 CHANGES IN EMPLOYMENT STATUS; TRANSFERS OF EMPLOYMENT: If a Participant ceases to be an Employee but continues in the employment of (i) the Company in some other capacity or (ii) a Related Corporation, he shall nevertheless continue as a Participant hereunder until his participation is otherwise terminated in accordance with the provisions of the Plan; provided, however, that such Participant shall not be eligible to continue making Participant Contributions hereunder; and, provided further, that such Participant shall not be eligible to receive allocations of Employer Contributions hereunder. If a person is transferred directly from employment (iii) with the Company in a capacity other than as an Employee or (iv) with a Related Corporation, to employment with the Company as an Employee, his service with the Company or other Related Corporation shall be included in determining his eligibility to participate in the Plan under Section 3.1.

ARTICLE IV

CONTRIBUTIONS

4.1 PARTICIPANT CONTRIBUTIONS:

- [a] Rate of Participant Contributions: As a condition of eligibility to share in Company Matching Contributions (if any), a Participant must elect to make Participant Contributions to this Plan of from one percent to 16 percent of his Compensation (in whole percentage increments).
- [b] Participant Contribution Elections: At the Participant's election, Participant Contributions may be made either on an after-tax basis (hereinafter "After-Tax Contributions") or on a pre-tax basis under a salary reduction agreement (hereinafter "Pre-Tax Contributions"). Elections must be (i) made in accordance with procedures prescribed by the Plan Administrator, (ii) made on the form provided by the Plan Administrator for such purpose, and (iii) filed with the Plan

-5-

Administrator such number of days prior to the election's effective date as the Plan Administrator shall prescribe. Elections to make Participant Contributions shall be prospective only. Participant Contributions will be paid into the Trust Fund by the Company monthly.

[c] Change of Participant Contributions: A Participant may change the rate of Participant Contributions prospectively but not retroactively by completing and timely filing, as prescribed by the Plan Administrator, the appropriate form with the Plan Administrator (on the form provided by the Plan Administrator for such purpose) during the month preceding the month in which the change is to become effective. The rate of After-Tax Contributions or Pre-Tax Contributions may be changed not more frequently than once in any month effective as of the first pay period in the month following the timely filing of the appropriate form with the Plan Administrator.

In the event it becomes necessary for Highly Compensated Employees to change the rate of Contributions, as set forth in Section 4.3, said Highly Compensated Employees shall be permitted to do so in a manner deemed to be administratively feasible. Moreover, in the event the Plan Administrator, or its designee, deems it necessary to require Highly Compensated Employees to change the rate of contributions because of the likelihood of exceeding the limitations set forth in Section 4.3[a], said Highly Compensated Employees shall be required to do so, after appropriate notice, in a manner deemed to be administratively feasible.

[d] Suspension of Participant Contributions: A Participant may suspend or resume After-Tax Contributions or Pre-Tax Contributions by completing and timely filing, as prescribed by the Plan Administrator, the appropriate form with the Plan Administrator during the month preceding the month in which the change is to become effective. After-Tax Contributions or Pre-Tax Contributions shall be suspended or resumed effective for the first pay period in the month following the timely filing of the appropriate form provided by the Plan Administrator for such purpose.

- [e] Limit on After-Tax Contributions: For each Plan Year, the total of a Participant's After-Tax Contributions to this and any other qualified plan maintained by the Company or any Related Corporation, when added to his Pre-Tax Contributions for the Plan Year, may not exceed 16 percent of the Participant's Compensation for the Plan Year.
- [f] Limit on Pre-Tax Contributions: For each Plan Year, the total of a Participant's Pre-Tax Contributions to this and any other qualified plan maintained by the Company or any Related Corporation, when added to his After-Tax Contributions for the Plan Year, may not exceed 16 percent of the Participant's Compensation for the Plan Year; provided, however, that notwithstanding the foregoing provisions of this subsection [g], the total of a Participant's Pre-Tax Contributions for a year may not exceed \$7,000 (as adjusted for cost-of-living by the Secretary of the Treasury pursuant to Section 402[g][5] of the Code as of each January 1).

-6-

- 4.2 COMPANY MATCHING CONTRIBUTIONS: For the Short Plan Year during which the Plan is adopted and each Plan Year thereafter, the Company shall contribute to the Plan such amounts, which amounts will be "Company Matching Contributions" hereunder, as follows:
 - [a] For the Short Plan Year, the Company shall contribute as Company Matching Contributions hereunder an amount equal to 66 percent of the Participants' Pre-Tax Contributions and After-Tax Contributions for such Short Plan Year not in excess of six percent of such Participants' Compensation for such Short Plan Year.
 - [b] For the Plan Year which begins January 1, 1995 and ends
 December 31, 1995, the Company shall contribute as Company
 Matching Contributions hereunder an amount equal to 66 percent
 of the Participants' Pre-Tax Contributions and After-Tax
 Contributions for such 1995 Plan Year not in excess of six
 percent of such Participants' Compensation for such 1995 Plan
 Year.
 - [c] For any Plan Year which begins after December 31,1995, the Company shall contribute as Company Matching Contributions hereunder an amount equal to 50 percent of the Participants' Pre-Tax Contributions and After-Tax Contributions for such Plan Year not in excess of six percent of such Participants' Compensation for such Plan Year.
 - [d] For the Short Plan Year or any other Plan Year, the Company may in its discretion contribute as additional Company Matching Contributions hereunder such amount (if any) as shall be determined by the board of directors of the Company.

Company Matching Contributions shall be paid to the Trustee within the period of time prescribed by law to permit a Federal income tax deduction for such year. Company Matching Contributions shall be (i) made on behalf of Employees of the Company who were Participants during the period for which such Company Matching Contributions were authorized to be made, (ii) made on account of such Participants' Pre-Tax Contributions and After-Tax Contributions that were authorized to be matched with respect to such period, and (iii) allocated to each such Participant's Account, in accordance with the provisions of Section 5.2, in proportion to each such Participant's Pre-Tax Contributions and After-Tax Contributions that were authorized to be matched with respect to such period. The Company that makes a Company Matching Contribution may, with the approval of the Plan Administrator, specify that such Company Matching Contribution be treated as a Pre-Tax Contribution for purposes of the tests described in Section 4.3, in accordance with Section 401(k)(3)(D)(ii) of the Code.

4.3 CONTRIBUTION LIMITATIONS:

[a] Limitations: Notwithstanding the provisions of Sections 4.1 and 4.2, Participant Contributions and Company Matching

Contributions to this Plan and any other qualified plan maintained by the Company or any Related Corporation are subject to the following limitations:

[1] The "Average Actual Deferral Percentage" (as defined in subsection [c][2] below) for "Highly Compensated Employees" (as defined in subsection [c][4] below) for each Plan Year must be no greater than 1.25 times the Average Actual Deferral

-7-

Percentage for "Non-Highly Compensated Employees" (as defined in subsection [c][4] below) for that Plan Year; or, alternatively, the Average Actual Deferral Percentage for Highly Compensated Employees may be two times the Average Actual Deferral Percentage for Non-Highly Compensated Employees if the Average Actual Deferral Percentage for Highly Compensated Employees is not more than two percentage points higher than the Average Actual Deferral Percentage for Non-Highly Compensated Employees for that Plan Year.

- [2] The "Average Actual Contribution Percentage" (as defined in subsection [c][1] below) for Highly Compensated Employees for each Plan Year must be no greater than 1.25 times the Average Actual Contribution Percentage for Non-Highly Compensated Employees for that Plan Year; or, alternatively, the Average Actual Contribution Percentage for Highly Compensated Employees may be two times the Average Actual Contribution Percentage for Non-Highly Compensated Employees if the Average Actual Contribution Percentage for Highly Compensated Employees is not more than two percentage points higher than the Average Actual Contribution Percentage for Non-Highly Compensated Employees for that Plan year.
- [3] Anything contained in this subsection [a] to the contrary notwithstanding, in computing the Average Actual Deferral Percentage and the Average Actual Contribution Percentage hereunder, such computations shall be made in accordance with Treasury Regulation Section 1.401(m)-2 to prevent the multiple use of the alternative limitations described in clauses [1] and [2] above.
- $\begin{tabular}{ll} [b] & {\tt Return \ of \ Excess \ Contributions \ and \ Excess \ Aggregate \ Contributions:} \end{tabular}$
 - If at the end of the Plan Year, or as soon as [1] administratively feasible, it is determined that the Pre-Tax Contributions made on behalf of Highly Compensated Employees would otherwise exceed the limitations of subsection [a][1] above, first unmatched and then matched Pre-Tax Contributions, and earnings attributable to such Pre-Tax Contributions, shall be returned to Highly Compensated Employees in the order of their "Actual Deferral Percentages" (as defined in subsection [c][2] below) beginning with those Highly Compensated Employees with the highest Actual Deferral Percentages. Alternatively, but only to the extent permitted in regulations promulgated by the Secretary of the Treasury, instead of receiving a distribution of "Excess Contributions" (as defined in subsection [c][6] below), a Highly Compensated Employee may elect to have the Excess Contribution treated as an amount distributed to the Highly Compensated Employee and then contributed to the Plan by such Highly Compensated Employee as an After-Tax Contribution, or the Company may recharacterize such amounts in its sole discretion. In addition, a Highly Compensated Employee may redirect Pre-Tax Contributions as After-Tax Contributions for the balance of the Plan Year if it is determined that the Pre-Tax Contributions on

behalf of such Highly Compensated Employee would otherwise exceed the limitations of subsection [a][1] above.

- [2] In the event After-Tax Contributions or Company Matching Contributions made by or on behalf of Highly Compensated Employees would otherwise exceed the limitations of subsection [a][2] above for a Plan Year, such After-Tax Contributions or Company Matching Contributions, and the earnings attributable to such After-Tax Contributions or Company Matching Contributions, as applicable, will be returned in the following order and assets shall be liquidated on a pro rata basis from all investment categories.
 - [A] excess unmatched After-Tax Contributions, and earnings attributable to such unmatched After-Tax Contributions, shall be returned to Highly Compensated Employees;
 - [B] excess matched After-Tax Contributions, and earnings attributable to such matched After-Tax Contributions, shall be returned to Highly Compensated Employees on the basis of the respective portions of the excess contributions attributable to each Highly Compensated Employee;
 - [C] excess Company Matching Contributions, and earnings attributable to such Company Matching Contributions, shall be distributed to each Highly Compensated Employee to whose Account "Excess Aggregate Contributions" (as defined in subsection [c][5] below) were allocated for the Plan Year.

[c] Definitions:

- [1] The "Average Actual Contribution Percentage" means the average (expressed as a percentage rounded to the nearest hundredth of a percentage point) of the "Actual Contribution Percentages" (as hereinafter defined) for a specified group of Employees for a Plan Year. The "Actual Contribution Percentage" for any Plan Year means the ratio (expressed as a percentage rounded to the nearest hundredth of a percentage point) of After-Tax Contributions and Company Matching Contributions (in the aggregate) made under the Plan or any other plan maintained by the Company or a Related Corporation by or on behalf of an Employee for the Plan Year to that Employee's "compensation" (as defined in clause [3] below) for the Plan Year.
- [2] The "Average Actual Deferral Percentage" means the average (expressed as a percentage rounded to the nearest hundredth of a percentage point) of the "Actual Deferral Percentages" (as hereinafter defined) for a specified group of Employees for a Plan Year. The "Actual Deferral Percentage" for any Plan Year means the ratio (expressed as a percentage rounded to the nearest hundredth of a percentage point) of Pre-Tax Contributions and Qualified Non-Elective Contributions made under the Plan or any other plan maintained by the Company or a Related Corporation on behalf of an Employee for the Plan Year to that Employee's "compensation" (as defined in clause [3]

-9-

below) for the Plan Year. Pre-Tax Contributions and Qualified Non-Elective Contributions shall be taken into account in determining the Actual Deferral Percentage for a Plan Year only if (i) the Pre-Tax Contributions and Qualified Non-Elective Contributions relate to compensation that either would have been received by the Employee in the Plan Year (but for the deferral election) or is attributable to services performed by the Employee in the Plan Year and would have been received by the Employee within two and one-half months after the close of the Plan Year (but for the deferral election), and (ii) are not contingent on participation or performance of services after such date and the

Pre-Tax Contribution or Qualified Non-Elective Contribution is actually paid to the Trust no later than 12 months after the Plan Year to which the Pre-Tax Contribution or Qualified Non-Elective Contribution relates.

- [3] For purposes of this Section 4.3, "compensation" means compensation for services performed for the Company or any Related Corporation that is currently includible in gross income plus any amount contributed to this Plan or any other plan maintained by the Company or any Related Corporation as a pre-tax contribution plan, including any amount deferred under a Code Section 125 plan maintained by the Company or any Related Corporation.
- [4] The term "Highly Compensated Employee" means "highly compensated active employees" and "highly compensated former employees" (as hereinafter defined).

A "highly compensated active employee" means any Employee who performs services for the Company during the "determination year" (as hereinafter defined) and who, during the "look-back year" (as hereinafter defined),: (i) received compensation from the Company in excess of \$75,000 (as adjusted for cost-of-living pursuant to Section 415(d) of the Code); (ii) received compensation from the Company in excess of \$50,000 (as adjusted for cost-of-living pursuant to Section 415(d) of the Code) and was a member of the top-paid group for such year; or (iii) was an officer of the Company and received compensation during such year that is greater than 50 percent of the dollar limitation in effect under Section 4 15(b)(1)(A) of the Code. The term "highly compensated active employee" also includes: (iv) Employees who are described in the preceding sentence if the term "determination year" is substituted for the term "look-back year" and are one of the 100 employees who received the most compensation from the Company during the determination year; and (v) Employees who are 5 percent owners at any time during the look-back year or the determination year. If no officer has satisfied the compensation requirement of item (iii) above during either a determination year or a look-back year, the highest-paid officer for such year shall be treated as a "highly compensated active employee".

-10-

For purposes of this clause [4], the "determination year" shall be the Plan Year; and the "look-back year" shall be the 12-month period immediately preceding the determination year.

A "highly compensated former employee" means any Employee who (A) separated from service (or was deemed to have separated from service) prior to the determination year, (B) performs no service for the Company during the determination year, and (C) was a highly compensated active employee for either the year of separation or any determination year ending on or after the Employee's 55th birthday.

If an Employee is, during a determination year or look-back year, a family member of either a 5 percent owner who is an active or former Employee or a Highly Compensated Employee who is one of the ten most Highly Compensated Employees ranked on the basis of compensation paid by the Company during such year, the family member shall be aggregated with the 5 percent owner or the top-ten Highly Compensated Employee, as applicable. In such case, the family member and 5 percent owner or top-ten Highly Compensated Employee, as applicable, shall be treated as a single Employee receiving compensation and Plan Compensation and contributions. For purposes of this clause [4], "family member" includes the spouse, lineal ascendants and descendants of the Employee or former Employee and the spouses of such lineal ascendants and descendants.

The determination of who is a Highly Compensated Employee, including determinations of the number and

identity of Employees in the top-paid group, the top 100 Employees, the number of Employees treated as officers and the compensation to be considered, will be made in accordance with Section $414\,(g)$ of the Code and the regulations promulgated thereunder.

The term "Non-Highly Compensated Employee" means any Employee who is not a Highly Compensated Employee.

- [5] "Excess Aggregate Contributions" means, with respect to any Plan Year, the excess of:
 - [A] the aggregate amount of After-Tax Contributions and Company Matching Contributions (and Qualified Non-Elective Contributions or elective contributions taken into account in computing the Average Actual Contribution Percentage) made on behalf of Highly Compensated Employees, over
 - [B] the maximum aggregate amount of After-Tax Contributions and Company Matching Contributions permitted under the Average Actual Contribution Percentage test described in subsection [a][1] above.
- [6] "Excess Contributions" means, with respect to any Plan Year, the excess of:

-11-

- [A] the aggregate amount of Pre-Tax Contributions actually paid over to the Trust on behalf of Highly Compensated Employees, over
- [B] the maximum amount of Pre-Tax Contributions permitted under the Average Actual Deferral Percentage test described in subsection [a][2] above.
- 4.4 QUALIFIED NON-ELECTIVE CONTRIBUTIONS: In lieu of distributing Excess Contributions as provided in Section 4.3[b][1], or Excess Aggregate Contributions as provided in Section 4.3[b][2], the Company may make "Qualified Non-Elective Contributions" (as hereinafter defined) on behalf of Non-Highly Compensated Employees that are sufficient to satisfy either the Average Actual Deferral Percentage test or the Average Actual Contribution Percentage test, or both, pursuant to regulations promulgated under the Code. In addition, the Company may make additional discretionary contributions on behalf of Non-Highly Compensated Employees that are sufficient to satisfy the Average Actual Contribution Percentage Test, pursuant to regulations promulgated under the Code.

"Qualified Non-Elective Contributions" mean contributions made by the Company (other than Company Matching Contributions) and allocated to Participants' Accounts that (i) Participants may not elect to receive in cash until distributed from the Plan, (ii) are nonforfeitable when made, and (iii) are distributable only in accordance with the distribution provisions applicable to Pre-Tax Contributions and Company Matching Contributions.

If the Company makes Qualified Non-Elective Contributions to the Plan, the amount of such Qualified Non-Elective Contributions for a Plan Year shall be an amount determined by the Company.

Allocation of Qualified Non-Elective Contributions shall be made (i) on a per capita basis or (ii) in the ratio which each Non-Highly Compensated Employee's "compensation" (as defined in Section 4.3[c][3] for the Plan Year bears to the total compensation of all Non-Highly Compensated Employees for such Plan Year; provided, however, that unless the Company determines otherwise, only Non-Highly Compensated Employees who are Participants shall be entitled to share in the allocation of Qualified Non-Elective Contributions.

4.5 LIMITATION ON ANNUAL ADDITIONS: For purposes of this Section 4.5, "annual additions" mean Company Contributions, Participant Contributions and forfeitures (if any). "Annual additions" shall not include (i) Rollover Contributions (if any), or (ii) company contributions described in Section 415(6) of the Code. For purposes of applying the limitations of Section 415 of the Code, the "limitation year" is the Plan Year. If the annual additions to the Account of any Participant, attributable to all defined contribution plans of the Company and any Related Corporation, would exceed either (A) \$30,000 (as adjusted for cost-of-living by the Secretary of the Treasury as of each January 1 for any limitation year ending during the calendar year) or (B) 25 percent of the Participant's "compensation" (as hereinafter defined), the excess amount

[a] First, unmatched Participant After-Tax Contributions, to the extent that the return would reduce the excess amount, will be returned to the Participant;

-12-

- [b] Second, matched Participant After-Tax Contributions, to the extent the return would reduce the excess amount, will be returned to the Participant with any Matching Company Contributions reduced as a consequence and allocated under subsection [d] or [e] below; and
- [c] Third, Company Contributions, to the extent the reduction would reduce the excess amount, will be allocated under subsection [d] or [e] below.
- [d] The amount of any excess attributable to Company Contributions will be applied to offset Company Matching Contributions for the limitation year; and
- [e] To the extent that the excess amounts described in subsection [d] above cannot be applied to offset Company Matching Contributions for the limitation year, the excess amounts will be allocated to a suspense account and applied to offset Company Matching Contributions in succeeding limitation years, as necessary.

For purposes of limiting annual additions under this Section 4.5 and limiting combined benefits and contributions under Section 4.6, "compensation" means wages for federal tax withholding purposes, as defined in Section 3401(a) of the Code, but determined without regard to any rules that limit remuneration included in wages based on the nature or location of employment or the services performed. Thus, for purposes of applying the limitations of Sections 4.5 and 4.6, compensation for a limitation year is the compensation actually paid or includible in gross income during such limitation year. Notwithstanding the preceding sentences, "compensation" for a participant in a defined contribution plan who is "permanently and totally disabled" (as defined in Section 22(e)(3) of the Code) is the compensation such participant would have received for the limitation year if the participant had been paid at the rate of compensation paid immediately before becoming permanently and totally disabled; provided, however, such imputed compensation for the disabled participant may be taken into account only if the participant is not a Highly Compensated Employee (as defined in Section 4.3[c][4]) and contributions made on behalf of such participant are nonforfeitable when made.

Anything contained herein to the contrary notwithstanding, compensation taken into account for purposes of applying the limitations of Sections 4.5 and 4.6 for any participant shall not exceed \$150,000\$ (subject to adjustment annually by the Secretary of the Treasury as provided in Sections 401(a)(17)(b) and 415(d) of the Code).

4.6 LIMITATION ON COMBINED BENEFITS AND CONTRIBUTIONS UNDER ALL DEFINED BENEFIT PLANS AND DEFINED CONTRIBUTION PLANS OF THE COMPANY AND ANY RELATED CORPORATION: In any limitation year, if the Company makes contributions to a defined benefit plan on behalf of an Employee who is also a Participant in this Plan, then the sum of the "defined benefit plan fraction" and the "defined contribution plan fraction" (as defined in subsections [a] and [b] below, as applicable) for the Employee for the limitation year may not exceed 1.0. In any limitation year, if the sum of the defined benefit plan fraction and the defined contribution plan fraction on behalf of a Participant would exceed 1.0, the benefit accrued on that Participant's behalf under the defined benefit plan will be limited, to the extent necessary, to prevent the sum of the defined benefit plan fraction and the defined contribution plan fraction from exceeding 1.0.

-13-

[a] Defined Benefit Plan Fraction: The "defined benefit plan fraction" is a fraction, the numerator of which is the projected annual benefit of the Participant under the defined benefit plan (determined as of the close of the limitation year) and the denominator of which is the lesser of the following amounts determined for that year and for each prior year of service with the Company:

- [1] the product of 1.25 times the maximum benefit dollar limitation in effect for the limitation year; or
- [2] the product of 1.4 times 100 percent of the Participant's average compensation for his high three consecutive calendar years.

In determining the defined benefit plan fraction, all defined benefit plans of the Company or a Related Corporation (whether or not terminated) shall be treated as one defined benefit plan.

- [b] Defined Contribution Plan Fraction: The "defined contribution plan fraction" is a fraction, the numerator of which is the sum of the annual additions to the Participant's accounts under all defined contribution plans of the Company or a Related Corporation as of the close of the limitation year and the denominator of which is the sum of the lesser of the following amounts determined for that year and for each prior year of service with the Company:
 - [1] the product of 1.25 times the dollar limitation in effect under Section 415(c)(1)(A) of the Code for the limitation year (without regard to Section 415(c)(6) of the Code); or
 - [2] the product of 1.4 times an amount equal to 25 percent of the Participant's compensation for the limitation year.
- 4.7 EXCLUSIVE BENEFIT: This Plan and Trust have been established for the exclusive benefit of Participants and their Beneficiaries. Under no circumstances may any funds contributed to the Plan or held by the Trustee at any time revert to or be used by the Company, except that all contributions by the Company to the Plan are conditioned upon both the qualification of the Plan and the deductibility of such contributions. At the Company's written request to the Trustee, a contribution by the Company shall be returned to the Company within one year of (i) the date of payment if the contribution was made by mistake of fact; (ii) the date initial qualification or qualification on Plan amendment is denied, provided the amendment is submitted to the Internal Revenue Service for a determination on qualification within one year after it is adopted; or (iii) the date deduction of a contribution is disallowed. The amount returned may not include net earnings but may be adjusted for net losses attributable to the contribution.
- 4.8 COMPANY'S OBLIGATIONS: The adoption and continuance of the Plan will not be deemed to constitute a contract between the Company and any Employee or Participant, nor to be consideration for, nor inducement nor condition of, the employment of any person. Nothing in this Plan will be deemed to give any Employee or Participant the right to be retained in the employ of the Company or to interfere with the right of the Company to discharge any Employee or Participant at any time, nor will it be deemed to give the Company the right to

-14-

require the Employee or Participant to remain in its employ, nor will it interfere with the right of any Employee or Participant to terminate his employment at any time.

ROLLOVER CONTRIBUTIONS: The Trustee shall accept funds transferred from a trust forming part of a plan that is qualified under Section 401(a) of the Code or from a conduit individual retirement account for the benefit of a Participant under this Plan if the trust from which the funds are so transferred allows the transfer; provided, however, that the Plan Administrator shall not permit the Trustee to accept a transfer of funds if such funds would require this Plan to pay benefits in the form of an annuity, or if, in the opinion of counsel, such funds would jeopardize the tax-qualified status of the Plan. Any amount transferred to the Plan in accordance with the provisions of this Section 4.9 shall be known as a "Rollover Contribution". In the event the Internal Revenue Service determines that all or any portion of a Rollover Contribution does not satisfy the requirements for rollover contributions under law, such Rollover Contribution, or the applicable portion thereof, shall be distributed to the Participant in a single sum as soon as administratively possible. The Trustee shall maintain a separate account for any Rollover Contributions made hereunder.

- 5.1 PARTICIPANT'S SUBACCOUNTS: The Plan Administrator will maintain one or more separate subaccounts for each Participant in the following categories to which contributions and Trust earnings or losses, as applicable, will be credited as provided in the Plan:
 - [a] Matched After-Tax Subaccount, to which a Participant's matched After-Tax Contributions (if any) and earnings and losses attributable thereto will be credited;
 - [b] Unmatched After-Tax Subaccount, to which a Participant's Unmatched After-Tax Contributions (if any) and earnings and losses attributable thereto will be credited;
 - [c] Matched Pre-Tax Subaccount, to which a Participant's matched Pre-Tax Contributions (if any) and earnings and losses attributable thereto will be credited;
 - [d] Unmatched Pre-Tax Subaccount, to which a Participant's unmatched Pre-Tax Contributions (if any) and earnings and losses attributable thereto will be credited;
 - [e] Company Contributions Subaccount, to which a Participant's share of Company Matching Contributions (if any) and earnings and losses attributable thereto will be credited;
 - [f] Rollover Subaccount, to which a Participant's Rollover Contributions (if any) and earnings and losses attributable thereto will be credited; and

-15-

[g] Transfer Subaccount, to which a Participant's Transfer Contributions (if any) and earnings and losses attributable thereto will be credited.

5.2 ALLOCATION OF CONTRIBUTIONS:

- [a] Allocation of Company Matching Contributions: Company Matching Contributions made pursuant to Section 4.2 shall be allocated as follows:
 - [1] Company Matching Contributions for the Short Plan Year, which Company Matching Contributions are made pursuant to Section 4.2[a], shall be allocated to Participants' Accounts, with the allocation to each such Participant's Account to equal 66 percent of such Participant's Pre-Tax Contributions and After-Tax Contributions for the Short Plan Year not in excess of six percent of such Participant's Compensation for such Short Plan Year.
 - [2] Company Matching Contributions for the Plan Year that begins January 1, 1995 and ends December 31, 1995, which Company Matching Contributions are made pursuant to Section 4.2[b], shall be allocated to Participants' Accounts, with the allocation to each such Participant's Account to equal 66 percent of such Participant's Pre-Tax Contributions and After-Tax Contributions for the 1995 Plan Year not in excess of six percent of such Participant's Compensation for such 1995 Plan Year.
 - [3] Company Matching Contributions for any Plan Year beginning after December 31, 1995, which Company Matching Contributions are made pursuant to Section 4.3[c], shall be allocated to Participants' Accounts, with the allocation to each such Participant's Account to equal 50 percent of such Participant's Pre-Tax Contributions and After-Tax Contributions for the Plan Year not in excess of six percent of such Participant's Compensation for such Plan Year.
 - [4] Discretionary Company Matching Contributions (if any) for the Short Plan Year or any other Plan Year, as applicable, which discretionary Company Matching Contributions are made pursuant to Section 4.3[d], shall be allocated to Participants' Accounts, with the allocation to each such Participant's Account to equal a specified percentage of such Participant's Pre-Tax Contributions and After-Tax Contributions for the Plan Year or Short Plan Year, as applicable, not in excess of a specified percentage of such Participant's Compensation for such Plan Year or Short Plan Year, as

applicable, and which percentages shall be specified by the board of directors of the Company.

[b] Allocation of Participant Contributions: Participant
Contributions received by the Trustee will be credited to the
Participant's Account as soon as the amounts to be credited to
such Participant's Account can be determined.

-16-

5.3 VESTING OF PARTICIPANT'S ACCOUNT:

- [a] Participant Contributions: A Participant's interest in his Participant Contributions will at all times be fully vested and nonforfeitable.
- [b] Company Contributions: A Participant's interest in his Company Contributions will at all times be fully vested and nonforfeitable.
- [c] Rollover Contributions: A Participant's interest in his
 Rollover Contributions will at all times be fully vested and
 nonforfeitable.
- [d] Transfer Contributions: A Participant's interest in his Transfer Contributions will at all times be fully vested and nonforfeitable.
- TRANSFER OF ACCOUNTS UPON TRANSFER OF EMPLOYMENT: Notwithstanding any other provision of the Plan to the contrary, the Plan Administrator may direct the Trustee to transfer to or accept a transfer from any other defined contribution plan maintained by the Company or a Related Corporation that is qualified under Section 401(a) of the Code of the account(s), if any, of an employee whose employment transfers (i) to employment covered by the Plan from employment covered by such other plan or (ii) from employment covered by the Plan to employment covered by such other plan. If the accounts(s) from another qualified defined contribution plan maintained by the Company or a Related Corporation are transferred to or merged into the Plan, such account(s) shall be held and administered in accordance with any restrictions applicable to them under such other plan to the extent required by law and shall be accounted for separately to the extent necessary to accomplish the foregoing. Under rules prescribed by the Plan Administrator, any applications, elections, designations and waivers under the transferor plan that are applicable to such transferred account(s) shall be applicable hereunder. In addition, any loan outstanding under the transferor plan with respect to such transferred account(s) shall become a loan under the Plan and shall continue to be repaid and otherwise governed under the Plan by the terms of such loan as made under the transferor plan. The Plan Administrator shall direct the Trustee to transfer any such account(s) at such time(s) as the Plan Administrator determines that the limitations contained in Section 4.3 and any other applicable limitations have been or will be satisfied and any remedial action required to comply with such limitations has been or will be taken under the Plan or the transferor plan, as applicable.

ARTICLE VI

INVESTMENT OF ACCOUNTS

- 6.1 INVESTMENT CATEGORIES: The following investment categories will be offered:
 - [a] T. Rowe Price Prime Reserve Fund;
 - [b] T. Rowe Price Stable Value Fund;
 - [c] T. Rowe Price Spectrum Income Fund;

-17-

- [d] T. Rowe Price Equity Index Fund;
- [e] T. Rowe Price Capital Appreciation Fund;
- [f] T. Rowe Price International Stock Fund;

- [q] T. Rowe Price New America Growth Fund;
- [h] Cyprus Stock Fund; and
- [i] Cliffs Stock Fund;

provided, however, that: (i) the Cliffs Stock Fund shall not be available as an investment category until the required registration has been made with the Securities and Exchange Commission; and (ii) the Cyprus Stock Fund shall only be available as an investment category for a period of 24 months, which 24-month period shall commence with the day upon which assets are first transferred from the Cyprus Amax Minerals Company Savings Plan & Trust (the "Cyprus Savings Plan") to this Plan (the "Transfer Date"); and, provided further, that a Participant may only invest such amounts in the Cyprus Stock Fund as are invested in the common stock of Cyprus Amax Minerals Company under the Cyprus Savings Plan on the day preceding the Transfer Date. In connection with item (ii) above, any Participant who maintains an investment in the Cyprus Stock Fund must liquidate such investment prior to the expiration of the 24-month period described above and reinvest such liquidated amount in one or more of the investment categories then available under the Plan.

- 6.2 CONTINUING INVESTMENT: A Participant may direct the continuing investment of Participant Contributions, Company Contributions, Rollover Contributions and Transfer Contributions in any one or more of the investment categories specified in Section 6.1 at any time, subject, however, to the limitations described in Section 6.1 regarding investment in the Cyprus Stock Fund; and such investment direction shall be effective for the month of the direction. If the Participant directs continuing investment in more than one investment category, the portion in each investment category must be specified as a percentage of the total in multiples of one percent. A Participant may change future investment directions at any time. This change will be effective for the month of the direction if the Trustee is notified of the change before the end of such month. Earnings on assets credited to a Participant's Account shall be invested in the investment category which produced such earnings.
- 6.3 CHANGE OF INVESTMENT HOLDINGS: A Participant may direct the Trustee on any business day to sell any investments in the Participant's account, and the Participant may direct that the proceeds of such sale be invested in any one or more of the other investment categories listed in Section 6.1, subject, however, to the limitations described in Section 6.1 regarding investment in the Cyprus Stock Fund.
- 6.4 INTERIM INVESTMENT: In the absence of effective investment directions, the Plan Administrator will direct the Trustee to invest any cash in the T. Rowe Price Prime Reserve Fund.
- 6.5 VALUATION OF PARTICIPANT'S ACCOUNT: Except as otherwise provided in the case of distributions under Article VII, a Participant's Account will be valued each business day that the Trustee is open for business.

-18-

ARTICLE VII

DISTRIBUTION FROM TRUST FUND

- 7.1 WHEN ACCOUNT BECOMES DISTRIBUTABLE: If a Participant dies, suffers Total Disability, retires or Terminates Employment for any other reason, his Account will be distributable.
- 7.2 TIME AND FORM OF PAYMENT:
 - Time of Payment: Upon Termination of Employment, a Participant may file a written claim for distribution of his Account with the Plan Administrator on a form prescribed by the Plan Administrator for such purpose. In such event, distribution of a Participant's Account will be made within 90 days after the calendar quarter during which the Participant's Termination of Employment occurs or the Participant receives his last payroll check. If no claim for distribution is filed by the Participant upon Termination of Employment, the Plan Administrator will schedule a payment date which shall be 60 days after the end of the Plan Year during which the Participant attains Normal Retirement Age; provided, however, that a Participant who Terminates Employment may elect a delayed distribution date, which delayed distribution date must be no later than April 1 of the calendar year following the calendar year in which the Participant attains age 70-1/2. A delayed distribution date may be accelerated at the

Participant's written direction to the Plan Administrator as follows: a Participant may file a written claim for distribution of his Account with the Plan Administrator, on a form prescribed by the Plan Administrator for such purpose, at any time after Termination of Employment; and, in such event, distribution of his Account will occur within 90 days following the calendar quarter during which the Participant files a written claim for such distribution.

- [b] Anything contained herein to the contrary notwithstanding distribution to a Participant must begin no later than April 1 of the calendar year following the calendar year in which the Participant attains age 70-1/2, with subsequent annual single sum payments made thereafter of amounts accrued (if any) during each subsequent calendar year. If the distribution of the entire interest of a Participant has begun in accordance with the foregoing provisions of this Section 7.2[b] and the Participant dies before his entire interest has been distributed to him, the remaining portion of such interest shall be distributed at least as rapidly as under the method of distribution being used at the date of the Participant's death. All distributions required under this Section 7.2[b] shall be determined and made in accordance with Treasury Regulations promulgated under Section 401(a)(9) of the Code, including the minimum distribution incidental benefit requirement of Treasury Regulation Section 1.401(a)(9)-2.
- [c] Payment in Cash or in Kind: At a Participant's or Beneficiary's request, payment of the portion of the Participant's Account invested in the Cliffs Stock Fund may be made in cash or in kind or partly in

-19-

cash and partly in kind; provided, however, that fractional shares of Cliffs Stock will be paid in cash. Payment of the remainder of the Participant's Account shall be made in cash. Distributions in cash will be the liquidation proceeds of the assets in the Participant's Account. The Plan Administrator will direct the Trustee to make payment in cash or in kind or partly in cash and partly in kind, as hereinbefore set forth; provided, however, that when directing the form of payment, the Plan Administrator will make any necessary adjustment on account of an outstanding loan the Participant may have under Section 7.9.

- [d] Form of Payment: Distribution of a Participant's Account will be made in a single sum payment.
- 7.3 DISPOSITION OF ACCOUNT ON TERMINATION OF EMPLOYMENT: Anything contained herein to the contrary notwithstanding, if a Participant's employment is terminated and the value of his account is \$3,500 or less (taking into account both Company Contributions and Participant Contributions) the Participant's Account shall be distributed as soon as administratively feasible following the Participant's Termination of Employment. If the value of the Participant's Account is in excess of \$3,500, such Account shall not be distributed to the Participant without his prior written consent and, if the Participant is married, the prior written consent of his spouse.
- 7.4 DIRECT ROLLOVER OF DISTRIBUTIONS: Notwithstanding any provision of the Plan to the contrary that would otherwise limit a "distributee's" (as defined in subsection [b] below) election under this Section 7.4, a distributee may elect, at the time and in the manner prescribed by the Plan Administrator, to have any portion of an "eligible rollover distribution" (as defined in subsection [d] below) paid directly to an "eligible retirement plan" (as defined in subsection [c] below) specified by the distributee in a "direct rollover" (as defined in subsection [a] below). For purposes of this Section 7.4, the following definitions shall apply:
 - [a] A "direct rollover" is a payment by the Plan to the eligible retirement plan specified by the distributee.
 - [b] A "distributee" includes an Employee or former Employee. In addition, the Employee's or former Employee's surviving spouse and the Employee's or former Employee's spouse or former spouse who is the alternate payee under a "qualified domestic relations order" (as defined in Section 414(p) of the Code) are "distributees" with regard to the interest of the spouse or former spouse.
 - [c] An "eligible retirement plan" is an individual retirement account described in Section 408(a) of the Code, an individual

retirement annuity described in Section 408(b) of the Code, an annuity plan described in Section 403(a) of the Code or a qualified trust described in Section 401(a) of the Code, that accepts the distributee's eligible rollover distribution. However, in the case of an eligible rollover distribution to the surviving spouse, an "eligible retirement plan" is an individual retirement account or individual retirement annuity.

-20-

- [d] An "eligible rollover distribution" is any distribution of all or any portion of the balance to the credit of the distributee, except that "eligible rollover distribution" does not include:

 (i) any distribution that is one of a series of substantially equal periodic payments (not less frequently than annually) made for the life (or life expectancy) of the distributee or the joint lives (or joint life expectancies) of the distributee and the distributee's designated beneficiary, or for a specified period of ten years or more; (ii) any distribution to the extent such distribution is required under Section 401(a)(9) of the Code; and (iii) the portion of any distribution that is not includible in gross income (determined without regard to the exclusion for net unrealized appreciation with respect to employer securities).
- 7.5 SPENDTHRIFT TRUST PROVISIONS: Except for benefits payable in accordance with the applicable requirements of a domestic relations order determined by the Plan Administrator to be a "qualified domestic relations order" (as defined in Section 414(p) of the Code), all benefits payable by the Plan will be paid only to the person entitled to them, and all payments will be made directly to that person and not to any other person or corporation. Benefits will not be subject to the claim of any creditor of a Participant, nor may payments be taken in execution by attachment or garnishment or by any other legal or equitable proceeding. No person will have any right to alienate, anticipate, commute, pledge, encumber or assign any benefits that he may expect to receive, contingently or otherwise, under this Plan, except the right to designate a Beneficiary or Beneficiaries.
- 7.6 DISTRIBUTION IN THE EVENT OF DEATH: If the Participant dies before distribution of his interest begins, distribution of the Participant's entire interest shall, at the Beneficiary's election, be made as soon as practicable following the Participant's death; provided, however, that if the Beneficiary elects a later distribution date, the Participant's entire interest must be distributed to the Beneficiary by December 31 of the calendar year containing the fifth anniversary of the Participant's death; and, provided further, that if the Beneficiary is the Participant's spouse and such spouse Beneficiary elects a later distribution date, the Participant's entire interest must be distributed to the Beneficiary no later than April 1 of the calendar year following the calendar year in which the Participant would have attained age 70-1/2.

7.7 BENEFICIARIES:

- [a] Designation of Beneficiary: A Participant shall designate a Beneficiary on a form prescribed by the Plan Administrator for such purpose and shall file such Beneficiary designation with the Plan Administrator. The Beneficiary designation will become effective only upon receipt of the form by the Plan Administrator, but upon receipt of the form the designation will relate back to and take effect as of the date the Participant signed the designation, whether or not the Participant is living at the time; provided, however, that neither the Plan Administrator nor the Company will be liable for any payment of the Participant's Account made prior to receipt of the form designating a Beneficiary.
- [b] Order of Payment: If the Participant fails to designate a
 Beneficiary before his death, or if no designated Beneficiary

-21-

survives the Participant, the Participant's undistributed Account will be paid in a single sum to the person, persons or entity in the first of the following classes of successive preference surviving the death of the Participant:

[1] spouse of the Participant,

- [2] children of the Participant,
- [3] parents of the Participant,
- [4] brothers and sisters of the Participant, or
- [5] the Participant's estate.

These provisions will be deemed modified where necessary to comply with the applicable law of any state.

- [c] Change of Beneficiary: A Participant may designate a new Beneficiary at any time by filing with the Plan Administrator a written change of Beneficiary on a form prescribed by the Plan Administrator for such purpose. The change will become effective only upon receipt of the form by the Plan Administrator, but upon receipt of the form the change will relate back to and take effect as of the date the Participant signed the request, whether or not the Participant is living at the time of receipt; provided, however, that neither the Plan Administrator nor the Company will be liable for any payment of the Participant's Account made prior to receipt of the form designating a change of Beneficiary.
- Spousal Consent: Anything contained in this Section 7.7 to the [d] contrary notwithstanding, no Beneficiary designation or change of Beneficiary under this Plan will be effective unless the spouse (if any) of the Participant consents in writing thereto. Such spousal consent shall be irrevocable and must acknowledge the effect of the election and be witnessed by a Plan representative or notary public. Spousal consent will not be required if it is established to the satisfaction of the Plan Administrator that such consent cannot be obtained because the Participant has no spouse or because the spouse cannot be located. Any election, consent or Beneficiary designation that has the effect of revoking a waiver of a surviving spouse's benefit may be made by a Participant without spousal consent anytime before payment of a Participant's Account is made under the Plan.
- [e] Uncertainty as to Right of Beneficiary: If the Plan Administrator is in doubt as to the right of any Beneficiary, the amount in question may be paid to the estate of the Participant, in which event neither the Plan Administrator nor the Company will be under any further liability to anyone with respect to the Account of the Participant.
- [f] Participant or Beneficiary Whose Whereabouts Are Unknown: In the case of any Participant or Beneficiary whose whereabouts are unknown, the Plan Administrator will notify the Participant or Beneficiary at his last known address by certified mail, return receipt requested, advising him of his right to a pending distribution. If the

-22-

Participant or Beneficiary cannot be located in this manner, the Plan Administrator may (i) direct the benefits be deposited with the registry of the appropriate district court, (ii) establish an account in the Participant's name until it is claimed or until proof of death satisfactory to the Plan Administrator is received by the Plan Administrator, or (iii) declare the Account forfeited; provided, however, that if a written claim for forfeited benefits is subsequently made by the Participant or Beneficiary, the amount forfeited, unadjusted for earnings or interest, will be paid to the Participant or Beneficiary, as applicable.

- 7.8 WITHDRAWALS DURING SERVICE: A Participant who is an Employee or who is in an ineligible class of employees and has not experienced a Termination of Employment may elect at any time, but not more than twice in any Plan Year, to withdraw, in whole or in part, the value of his Account, subject to the limitations set forth in this Section 7.8. A Participant shall initiate a withdrawal request by timely executing and filing a written election with the Plan Administrator on a form prescribed by the Plan Administrator for such purpose, which form shall require such information as the Plan Administrator from time to time determines is needed to process the withdrawal request. Any amounts withdrawn pursuant to this Section 7.8 may not be repaid.
 - [a] Order of Withdrawals: Withdrawals by a Participant shall be made from the Participant's subaccounts in the order listed

below:

- [1] any portion of his subaccounts attributable to the principal amount of the Participant's After-Tax Contributions (including the Participant's Transfer Contributions attributable to after-tax contributions) and income attributable thereto;
- [2] any portion of his subaccounts attributable to the principal amount of Rollover Contributions (including the Participant's Transfer Contributions attributable to rollover contributions), and then any portion of his subaccounts attributable to income on such Rollover Contributions (including income on the Participant's Transfer Contributions attributable to rollover contributions);
- in the case of a Participant who has attained at least age 59-1/2, any portion of his subaccounts attributable to the principal amount of Pre-Tax Contributions (including the Participant's Transfer Contributions attributable to pre-tax contributions), and then any portion of his subaccounts attributable to income on such Pre-Tax Contributions (including income on the Participant's Transfer Contributions attributable to pre-tax contributions); and
- [4] in the case of a Participant who has not attained age 59-1/2, any portion of his subaccounts attributable to the principal amount of Pre-Tax Contributions (including the Participant's Transfer Contributions attributable to pre-tax contributions).

No withdrawal may be made under clauses [1] through [4] above unless all amounts that may be withdrawn under each preceding clause have been withdrawn.

-23-

- [b] Restriction on Withdrawals of Pre-Tax Contributions Where Participant Is Less than Age 59-1/2: Anything contained in this Section 7.8 to the contrary notwithstanding, a Participant under the age of 59-1/2 may only withdraw the portion of his Account attributable to Pre-Tax Contributions "on account of hardship", as determined by the Plan Administrator on a uniform and nondiscriminatory basis in accordance with the following provisions:
 - [1] A withdrawal "on account of hardship" shall mean a distribution that is (i) made on account of a Participant's "immediate and heavy financial need" (as defined in clause [2] below), and (ii) "necessary to satisfy the immediate and heavy financial need" (as defined in clause [3] below).
 - [2] A distribution that is made on account of a Participant's "immediate and heavy financial need" means a distribution that is made for one or more of the following reasons:
 - [A] expenses for medical care described in Section 213(d) of the Code previously incurred by the Participant, the Participant's spouse or any "dependents" (as defined in Section 152 of the Code) of the Participant or necessary for those persons to obtain the medical care described in Section 213(d) of the Code;
 - [B] costs directly related to the purchase of a principal residence for the Participant (excluding mortgage payments);
 - [C] payment of tuition and related education fees for the next semester, quarter or 12 months of post-secondary education for the Participant or the Participant's spouse, children or "dependents" (as defined in Section 152 of the Code);
 - [D] payments necessary to prevent the eviction of the Participant from his principal residence or

foreclosure on the mortgage on that residence; and

- [E] any other financial need that the Commissioner of Internal Revenue, through the publication of revenue rulings, notices and other documents of general applicability, may from time to time designate as a deemed immediate and heavy financial need as provided in Treasury Regulation Section 1.401(k)-1(d)(2)(ii)(iv).
- [3] A distribution is "necessary to satisfy the immediate and heavy financial need" if:
 - [A] the distribution is not in excess of the amount of the immediate and heavy financial need of the Participant; provided, however, that the amount of an immediate and heavy financial need may include any amounts necessary to pay federal, state or local income taxes or penalties reasonably anticipated to result from the distribution; and

-24-

- [B] the Participant has obtained all distributions, other than hardship distributions, and all nontaxable (at the time of the loan) loans currently available under this Plan and all other plans maintained by the Company or a Related Corporation.
- [4] In the event a Participant receives a hardship withdrawal hereunder, he must suspend making Pre-Tax Contributions under the Plan (and under all other plans maintained by the Company or a Related Corporation) for the 12-month period commencing as of the date of such hardship withdrawal. Furthermore, after becoming eligible to recommence making Pre-Tax Contributions (i.e., following expiration of the above 12-month period), the maximum amount of Pre-Tax Contributions the Participant may contribute under the Plan (and all other plans maintained by the Company or a Related Corporation) may not, for the taxable year immediately following the taxable year in which the hardship withdrawal was made pursuant to this Section 7.8, exceed \$7,000 (as adjusted for cost-of-living by the Secretary of the Treasury pursuant to Section 402(g)(5) of the Code) reduced by the Pre-Tax Contributions made by the Participant under the Plan (and all other plans maintained by the Company or a Related Corporation) during the taxable year in which the hardship withdrawal was made.
- [c] Form of Payment: Any amount withdrawn under this Section 7.8 will be paid in a single sum.
- [d] Order of Asset Conversion: The Plan Administrator will convert assets to obtain withdrawal proceeds in a uniform order which may change from time to time. Participants shall be given notice of such order and of any changes.
- [e] Qualified Domestic Relations Order. In the event a domestic relations order is determined to be a "qualified domestic relations order" (as defined in Section 414(p) of the Code), any and all benefits assigned to an "alternate payee" (as defined in Section 414(p) of the Code) under such qualified domestic relations order shall be available for distribution to such alternate payee as soon as administratively feasible after the date specified in the qualified domestic relations order.

7.9 PARTICIPANT LOANS:

- [a] Administration: The Plan Administrator or its designee shall be responsible for administration of the loan program.
- [b] Eligibility: A Participant who is an Employee may borrow from his Account in the Plan. Except in the case of parties-ininterest, only active Employees are eligible to apply for a loan from the Plan.

[c] Conditions of Loan: Any loan from the Plan will be subject to the following conditions:

-25-

- [1] The loan will be in the form of cash and the Participant's Account will serve as the sole collateral for the loan.
- [2] The loan may not exceed the lesser of 50 percent of the market value of the Participant's Account at the time the Participant's written loan request is received by the Plan Administrator or \$50,000 reduced by the excess (if any) of the highest balance of loans outstanding during the one-year period ending on the date the new loan is made over the balance of loans outstanding on the date the new loan is made.
- [3] The minimum loan will be \$1,000.
- [4] The term of a loan will be not less than six months and not more than five-years (ten years in the case of a loan used to acquire a dwelling which will, within a reasonable time, be used as the Participant's principle residence); provided, however, that the repayment term of the loan shall be in increments of six months.
- [5] The Participant must execute and deliver to the Plan Administrator a payroll deduction authorization in a form satisfactory to the Plan Administrator for the term of the loan repayment.
- [6] The Participant must execute and deliver to the Plan Administrator a promissory note in a form satisfactory to the Plan Administrator.
- [7] One new loan will be allowed each Plan Year, but no more than one outstanding loan will be permitted at any time.
- [8] The loan will be treated as an asset of the Participant's Account.
- [9] Except in the case of parties-in-interest, only active Employees, excluding those Employees who are receiving long-term disability benefits or who are on a leave of absence from the Company, are eligible to apply for a loan from the Plan.
- [10] The limits of this Section 7.9 [c] will apply to all loans made to the Participant under all plans maintained by the Company or any Related Corporation.
- [d] Employee Direction for Loan Proceeds and Valuation of Accounts:
 - [1] Employee Direction: A Participant shall execute a written directive to the Plan Administrator setting forth the amount of the loan requested on a form prescribed by the Plan Administrator for such purpose.
 - [2] Valuation of Account: The valuation of an Account, or the sale or redemption of securities credited to an Account, or both, will be made on the date or dates selected by the Plan Administrator, and within a reasonable time after receipt by the Plan Administrator of a written application for a loan.

-26-

[3] Order of Subaccount and Asset Conversion: The Plan Administrator will convert assets in a Participant's subaccounts on a pro rata basis to obtain the loan proceeds in the following order: Unmatched After-Tax Subaccount, Matched After-Tax Subaccount, Rollover Subaccount, Transfer Subaccount, Unmatched Pre-Tax Subaccount, Matched Pre-Tax Subaccount and Company Contributions Subaccount.

- [e] Interest on Loan: Interest charged on loans will be a commercially reasonable rate published for the first business day of the month immediately preceding the month in which a written directive for a loan is received by the Plan Administrator.
- [f] Allocation of Loan Payments: Loan payments, which must be made at least quarterly, will be credited to a Participant's subaccounts in the reverse order from which such subaccounts provided the loan proceeds.
- [g] Prepayment of Loan:
 - [1] Prepayment Prior to Termination of Employment: At any time during a 12-month period and prior to a Participant's Termination of Employment, a Participant may make a single sum prepayment of all of an outstanding loan balance.
 - [2] Prepayment on Termination of Employment: If a loan is not prepaid prior to a Participant's Termination of Employment, the outstanding balance will constitute a distribution upon such Participant's Termination of Employment. If a loan is not prepaid prior to a Participant's Termination of Employment and the Participant is a "party-in-interest" (as defined in Section 3(14) of ERISA), the outstanding balance will be paid in monthly installments as described in subsection [h] below. Participants who do not experience a Termination of Employment but who are in an ineligible class of employees will continue to pay the outstanding loan balance in monthly installments.
- [h] Loans to Terminated Employees: Terminated employees who are parties-in-interest and who do not request a distribution of their Accounts will be permitted to continue to make payments on pre-existing loans if said terminated employees meet the criteria for the continuance of such loans established by the Plan Administrator. Deferred vested employees who are parties in interest shall be permitted to obtain a loan from the Plan if said individuals meet the criteria established by the Plan Administrator.
- [i] Loan Default: Any loan to an active employee will be treated as being in default if a scheduled monthly payment is not made within 90 days of the payment due date. Any loan to a terminated employee who is a party-in-interest will be treated as being in default if a scheduled monthly payment is not made within 30 days of the payment due date. A loan which is in default will be treated as a distribution.

-27-

ARTICLE VIII

FIDUCIARY OBLIGATIONS

- 8.1 GENERAL FIDUCIARY DUTIES: A fiduciary must discharge his duties under the Plan solely in the interest of Participants and Beneficiaries and for the exclusive purpose of providing benefits to Participants and their Beneficiaries and defraying reasonable expenses of administering the Plan. All fiduciaries must act with the care, skill, prudence and diligence under the circumstances then prevailing that a prudent man acting in a like capacity and familiar with such matters would use in the conduct of an enterprise of a like character and with like aims. Except as authorized by regulations of the Secretary of Labor, no fiduciary may maintain the indicia of ownership of any assets of the Plan outside the jurisdiction of the district courts of the United States. A fiduciary must act in accordance with the documents and instruments governing the Plan to the extent the documents and instruments are consistent with the requirements of law.
- 8.2 ALLOCATION OF FIDUCIARY RESPONSIBILITY: A named fiduciary may designate persons other than named fiduciaries to carry out fiduciary responsibilities under the Plan; provided, however, that fiduciary responsibilities to manage or control Plan assets may not be delegated except by appointment of an investment manager. Named fiduciaries are the Company and the Plan Administrator.
 - COMPENSATION AND EXPENSES OF FIDUCIARIES:

- [a] General Rules: A fiduciary is entitled to reasonable compensation for services rendered and to reimbursement for expenses properly and actually incurred in the performance of his duties under the Plan. However, a fiduciary who already receives full-time pay from the Company may receive no compensation from the Plan, except for the reimbursement of expenses properly and actually incurred. All compensation and expenses will be paid by the Plan, unless the Company elects to pay all or any part of the compensation and expenses.
- [b] Compensation of Trustee: The Trustee is entitled to such reasonable compensation for its services as the Company and the Trustee mutually shall determine.
- [c] Compensation of Persons Retained or Employed by Named Fiduciary: The compensation of all agents, counsel or other persons retained or employed by a named fiduciary will be determined by the named fiduciary employing such person.

ARTICLE IX

PLAN ADMINISTRATOR

9.1 APPOINTMENT OF PLAN ADMINISTRATOR: The Board of Directors will appoint a Plan Administrator who will hold office until resignation, death or removal by the Board of Directors. If the Board of Directors fails to appoint a Plan

-28-

Administrator, the Company will be the Plan Administrator. Any person may serve in more than one fiduciary capacity. Any group of persons may serve in the capacity of Plan Administrator. A Plan Administrator may resign at any time by giving written notice to the Board of Directors. At any time a Plan Administrator may be removed by the Board of Directors with or without cause. As soon as practicable following the death, resignation or removal of any Plan Administrator, the Board of Directors will appoint a successor. Written notice of appointment of a successor Plan Administrator will be given by the Board of Directors to the Trustee. Until receipt by the Trustee of written notice, the Trustee will not be charged with knowledge or notice of the change.

- 9.2 INFORMATION TO BE MADE AVAILABLE TO PLAN ADMINISTRATOR: To enable the Plan Administrator to perform all of its duties under the Plan, the Board of Directors will provide the Plan Administrator with access to any information the Plan Administrator requires. If required information is not available from the Company's records, the Plan Administrator may obtain the information from the Participants. The Plan Administrator and the Company may rely on and will not be liable for any information that an Employee provides either directly or indirectly.
- 9.3 DUTIES AND POWERS OF PLAN ADMINISTRATOR:
 - [a] General: The Plan Administrator shall have all the powers and authority as may be necessary to carry out the provisions of the Plan, including the discretionary power and authority to interpret and construe the Plan and to resolve any disputes arising thereunder, and the powers and authority expressly conferred upon it herein. Without limiting the generality of the foregoing, the Plan Administrator will decide all questions arising in the administration, interpretation and application of the Plan, including all questions relating to eligibility and distributions, except as may be reserved under this Plan to the Company. The decisions of the Plan Administrator will be final. The Plan Administrator will direct the Trustee concerning payments to be made out of Plan assets. All notices, directions, information and other communications to and from the Plan Administrator will be in writing.
 - [b] Employment of Advisers and Persons to Carry Out
 Responsibilities: The Plan Administrator may employ one or more
 persons to render advice with regard to any responsibility the
 Plan Administrator has under the Plan and may employ one or
 more persons (including the Trustee) to carry out any of its
 responsibilities under the Plan.
 - [c] Reporting and Disclosure: The Plan Administrator will be responsible for all applicable reporting and disclosure requirements of law. In connection therewith, the Plan Administrator will prepare, file with the Department of Labor or Internal Revenue Service, and furnish to Plan Participants

and Beneficiaries, when applicable, the following:

- [1] summary plan descriptions;
- [2] descriptions of modifications and changes;
- [3] annual reports;

-29-

- [4] terminal and supplementary reports;
- [5] registration statements; and
- [6] any other returns, reports or documents required by law.
- [d] Inspection of Documents: The Plan Administrator shall make available for inspection copies of the Plan, the latest annual report and agreements under which the Plan was established or is operated. The documents will be available for examination by any Participant or Beneficiary in the principal office of the Plan Administrator and in any other places necessary to make available all pertinent information to Participants and Beneficiaries. On written request by any Participant or Beneficiary, the Plan Administrator shall furnish a copy of the latest updated summary plan description, the latest annual report, any termination report and any agreements under which the Plan is established or operated.
- [e] Keeping of Records: The Plan Administrator will keep any records that are necessary or advisable in its judgment for the administration of this Plan.
- [f] Bonding of Fiduciaries and Plan Officials: The Plan Administrator shall procure bonds for every fiduciary of the Plan and for every person who handles funds or other property of the Plan that is not exempt from the bonding requirements. Bonds must conform to the requirements of law and will be in an amount of not less than \$1,000 or, if greater, ten percent of the amount of funds handled, with a generally applicable maximum of \$500,000.
- 9.4 NOTICES FROM PARTICIPANTS: Whenever provision is made in the Plan that a Participant may exercise an option or election or designate a Beneficiary, the action of the Participant will be evidenced in writing on forms provided by the Plan Administrator for such purpose, which forms shall be signed by the Participant and delivered to the Plan Administrator in person or by mail. Written notice will not be effective until received by the Plan Administrator.

9.5 CLAIMS PROCEDURES:

Filing and Initial Determination of Claim: Any Participant or Beneficiary (the "claimant") or his duly authorized representative may file a claim for a Plan benefit to which the claimant believes he is entitled. A claim must be in writing and delivered to the Plan Administrator in person or by certified mail, postage prepaid. Within 90 days after receipt of a claim, the $\overline{\text{Plan}}$ $\overline{\text{Administrator}}$ will send the claimant by certified mail, postage prepaid, notice of the granting or denying, in whole or in part, of the claim, unless special circumstances require an extension of time for processing the claim; provided, however, that in no event may the extension exceed 90 days from the end of the initial 90-day period. If an extension of time is necessary, the claimant will receive a written notice prior to the expiration of the initial 90-day period regarding the need for an extension of time. The Plan Administrator has full discretion to deny or grant a claim in whole or in part. If notice

- [b] Duty of Plan Administrator Upon Denial of Claim: The Plan Administrator will provide a claimant who is denied a claim for benefits with a written notice setting forth, in a manner calculated to be understood by the claimant:
 - [1] the specific reason or reasons for the denial;
 - [2] specific references to pertinent Plan provisions upon which the denial is based;
 - [3] a description of any additional material or information necessary for the claimant to perfect the claim and an explanation of why the material is necessary; and
 - [4] an explanation of the Plan's claims' review procedure.
- Review of Claim Denial: Within 60 days after claimant's receipt [c] of written notice of a denial in whole or in part of his claim, the claimant or his duly authorized representative, by written application to the Plan Administrator in person or by certified mail, postage prepaid, may request a review of the denial, review pertinent documents and submit issues and comments to the Plan Administrator in writing. On receipt of a request for review, the Plan Administrator will render a decision, which decision will be written in a manner calculated to be understood by the claimant and will include reasons for the decision and specific references to pertinent Plan provisions upon which the decision is based. The decision on review will be made no later than 60 days after the Plan Administrator's receipt of a request for review, unless special circumstances require an extension of time for processing, in which case a decision will be rendered no later than 120 days after receipt of a request for review. If an extension of time is necessary, the claimant will be given written notice of the extension prior to expiration of the 60-day period. If notice of the decision on the review is not furnished under this subsection [c], the claim will be deemed denied.

ARTICLE X

TRUST FUND

10.1 TRUST FUND: The assets of the Plan shall be held in trust under the terms of the Trust Agreement between the Company and the Trustee, which Trust Agreement may be a trust established by the Company or a Related Corporation for the purpose of investing the assets of other plans qualified under Section 401(a) of the Code that are maintained by the Company or a Related Corporation; provided, however, that such trust is adopted as a part of this Plan; and, provided further, that the Trustee shall maintain separate accounting records to reflect the share of the Trust Fund that is to be credited to this Plan.

-31-

10.2 INVESTMENT OF TRUST FUND:

- [a] General: The assets of the Plan shall be invested in accordance with the terms of the Plan and the Trust Agreement under which the assets are held.
- [b] Options, Rights and Warrants: In the event that any options, rights or warrants are granted or issued with respect to stock other than Cliffs Stock held by the Trustee for a Participant, the Trustee shall be directed to sell such options, rights or warrants if there is a market for them. The net cash proceeds from the sale will be credited ratably to the Account of each Participant concerned. If the Trustee receives stock as a dividend or as the result of a stock split, the shares allocable to securities held in a Participant's Account will be added to his Account on a full or fractional share basis.
- [c] Voting: Cliffs Stock held by the Trustee for Participants and Beneficiaries with respect to which the Trustee receives written instructions from such Participants and Beneficiaries shall (i) be voted by the Trustee as directed by the Participant or Beneficiary, or (ii) not be voted by the Trustee if so directed by any Participant or Beneficiary. With respect to any Cliffs Stock allocated to a Participant's Account, at the time of a mailing to stockholders of the notice of any stockholders' meeting of Cleveland-Cliffs Inc, Cleveland-Cliffs

Inc, in conjunction with the Trustee, shall use its reasonable best efforts to cause to be delivered to each Participant and Beneficiary such notices and informational statements as are furnished to Cleveland- Cliffs Inc stockholders in respect of the exercise of voting rights, together with forms upon which the Participant or Beneficiary may confidentially instruct the Trustee, or revoke such instruction, with respect to the voting of shares of each class or series of Cliffs Stock allocated to his account. Upon timely receipt of directions, the Trustee shall vote each class or series of Cliffs Stock (with voting rights) allocated to a Participant's or Beneficiary's Account as directed by the Participant or Beneficiary. The Trustee shall vote or not vote each class or series of Cliffs Stock (with voting rights) that is not allocated to a Participant's or Beneficiary's Account and each class or series of Cliffs Stock (with voting rights) allocated to a Participant's or Beneficiary's Account which is not voted by the Participant or Beneficiary because the Participant or Beneficiary has not directed (or has not timely directed) the Trustee as to the manner in which such Cliffs Stock is to be voted, in the same proportion as those shares of the same class or series of Cliffs Stock for which the Trustee has received proper direction on such matter.

[d] Tender Offers for Cliffs Stock: Notwithstanding any other provision of the Plan to the contrary, if there is a tender offer for, or a request or invitation for tenders of, shares of a class or series of Cliffs Stock or other securities of the Company (or any Related Corporation) held by the Trustee for Participants or Beneficiaries (i) the Plan Administrator shall furnish to the Trustee, who shall then furnish to each Participant or Beneficiary, prompt notice of such tender offer for, or request or invitation for tenders of, such

-32-

shares or series of Cliffs Stock or other securities of the Company (or any Related Corporation), and (ii) the Trustee shall request from each Participant or Beneficiary instructions as to the tendering of each class or series of such shares of Cliffs Stock or other securities of the Company (or any Related Corporation) allocated to the Participant's or Beneficiary's Account. The Trustee shall tender only such shares of each class or series of Cliffs Stock or other securities of the Company (or any Related Corporation) for which the Trustee has received (within the time specified in the notification) tender instructions. With respect to shares of the class or series of Cliffs Stock or other securities of the Company (or any Related Corporation) which are not allocated to a Participant's or Beneficiary's Account, the Trustee shall tender such shares or series of Cliffs Stock or other securities of the Company (or any Related Corporation) in the same proportion as the number of such shares or series of Cliffs Stock or other securities of the Company (or any Related Corporation) for which instructions to tender are received bears to the total number of such shares or series of Cliffs Stock or other securities of the Company (or any Related Corporation) for which instructions from Participants or Beneficiaries could have been received. The Trustee shall not tender all other shares or series of Cliffs Stock or other securities of the Company (or any Related Corporation).

[e] Confidentiality: All instructions received by the Trustee from Participants or Beneficiaries pursuant to subsections [c] or [d] above shall be held by the Trustee in strict confidence and shall not be divulged to any person, including employees, officers and directors of the Company (or any Related Corporation); provided, however, that to the extent necessary for the operation of the Plan, such instructions may be relayed by the Trustee to a recordkeeper, auditor or other person providing services to the Plan if such person (i) is not the Company (or any Related Corporation) or any employee, officer or director thereof, and (ii) agrees not to divulge such directions to any other person, including employees, officers and directors of the Company (or any Related Corporation).

11.1 TERMINATION OF PLAN: Continuation of the Plan is not assumed as a contractual obligation by the Company, and the right is reserved to the Company, by action of the Board of Directors, to terminate this Plan and/or Trust in whole or in part at any time. Upon termination of the Plan, the Board of Directors shall have discretion to liquidate the Plan or continue a frozen plan making distributions as provided in Article VII. Termination of the Plan will in no event have the effect of revesting any part of the Trust Fund in the Company. Notice of termination will be given to the Trustee in the form of an instrument in writing executed by the Company pursuant to an action of its Board of Directors, together with a certified copy of the resolution of its Board of Directors to that effect. Termination of the Plan will take effect as of the

-33-

date specified by the Board of Directors. The Plan created by execution of this agreement will be terminated automatically in the event of the dissolution, consolidation or merger of the Company, or the sale by the Company of substantially all of its assets, if the resulting successor corporation or business entity does not continue the Plan. The Plan Administrator will file any termination reports required by law.

- 11.2 MERGER, CONSOLIDATION OR TRANSFER OF ASSETS OR LIABILITIES OF THE PLAN: The Board of Directors may merge or consolidate this Plan with any other plan may transfer the assets or liabilities of this Plan to any other plan or transfer the assets and liabilities of another plan to this Plan if each Participant in the Plan (if the Plan then terminated) would receive a benefit immediately after the merger, consolidation or transfer that is equal to or greater than the benefit he would have been entitled to receive immediately before the merger, consolidation or transfer (if the Plan had then terminated). If any merger, consolidation or transfer of assets or liabilities occurs, the Plan Administrator will file any reports required by law.
- AMENDMENTS TO PLAN: The Board of Directors may amend this Plan at any time; provided, however, that no amendment may cause the Trust Fund to be diverted to purposes other than for the exclusive benefit of Participants and their Beneficiaries. To the extent authorized by the Board of Directors, the Plan may be amended at any time and from time to time. No amendment may decrease the vested interest of any Participant. If an amended vesting schedule is adopted, any Participant who has three or more years of service at the later of the date the amendment is adopted or becomes effective may elect to remain under the Plan's prior vesting schedule. Notwithstanding the preceding sentence, no election need be provided for any Participant where the nonforfeitable percentage under the Plan as amended cannot be less than the percentage determined without regard to the amendment. The election must be made in writing, in such form and within such period of time prescribed by the Plan Administrator, in accordance with applicable regulations, and must be timely filed with the Plan Administrator. No amendment may discriminate in favor of Employees who are officers, shareholders or Highly Compensated Employees. Notwithstanding anything in this Plan to the contrary, the Plan may be amended at any time to conform to the provisions and requirements of federal and state laws or regulations or rulings issued pursuant to such federal and state laws. No such amendment will be considered prejudicial to the interest of any Participant or Beneficiary under this Plan.

11.4 PARTICIPATING COMPANIES:

[a] Requirements for Participation: Any Affiliated Company may participate in the Plan as a participating Company, provided its participation is approved by the Board of Directors and it executes an instrument of participation. The instrument of participation may contain terms and conditions approved by the Board of Directors and applicable only to the participating Company, and, to the extent qualification of the Plan is not affected thereby, will constitute an amendment of the Plan as it applies to the participating Company. To the extent there are no differences in Plan terms and conditions between the Plan as established and maintained by the Company and the Plan as adopted and maintained by a participating Company, such participating Company may indicate its participation in the Plan by executing the Plan document and any amendments thereto, which

documents and amendments (if any) shall constitute an instrument of participation hereunder.

- [b] Employee Transfers Within the Participating Group: A transfer of employment from one participating Company to another participating Company will not be considered a Termination of Employment and the Account of the Participant whose employment is so transferred will be transferred to the group of Accounts of the transferee participating Company. A transferee participating Company will continue any funding of the Plan for any Employee it acquires by transfer and credit the prior years of service of the transferred Participant.
- [c] Withdrawal by Participating Company: Any participating Company, by action of its board of directors, may withdraw from participation in the Plan upon written notice to the Plan Administrator and the Trustee, and may (i) treat the withdrawal as a termination of the Plan with respect to its Employees; (ii) treat the withdrawal as an amendment of the Plan and direct the Trustee to segregate and transfer to a separate trust and qualified plan the Accounts of Participants and Beneficiaries allocable to the participating Company; or (iii) treat the withdrawal as an amendment of the Plan and direct the Trustee to segregate and transfer the Accounts of Participants and Beneficiaries allocable to the Participating Company to another qualified plan in which the participating Company may participate; provided, however, that the value of the Account of any Participant immediately after the transfer will equal the value of his Account immediately before the transfer.
- [d] Contributions made on Behalf of Participating Company: In the event a participating Company is unable to make all or a portion of its Company Contribution by reason of not having sufficient current or accumulated profits to make such Company Contributions for a year in which a consolidated federal income tax return of the affiliated group is filed, another participating Company in the affiliated group, to the extent it has sufficient current or accumulated profits, may in its discretion make, on behalf of the participating Company, the portion of the Company Contributions the participating Company is otherwise prevented from making for the Plan Year.

ARTICLE XII

MISCELLANEOUS

- 12.1 BENEFITS TO BE PROVIDED SOLELY FROM THE TRUST FUND: All benefits payable under this Plan will be or provided solely from the Trust Fund, and the Company assumes no liability or responsibility for payment of benefits.
- 12.2 TEXT TO CONTROL: The headings of Articles, Sections, subsections and clauses are included solely for convenience of reference. If any conflict between any heading and the text of this Plan exists, the text will control.
- 12.3 SEVERABILITY: If any provision of this Plan is illegal and invalid for any reason, the illegality or invalidity will not affect the remaining provisions.

-35-

On the contrary, the remaining provisions will be fully severable, and this Plan will be construed and enforced as if the illegal or invalid provisions never had been inserted in the document.

- 12.4 JURISDICTION: This Plan will be construed and administered under the laws of the state of Delaware when the laws of that jurisdiction are not in conflict with federal law.
- 12.5 PLAN FOR EXCLUSIVE BENEFIT OF PARTICIPANTS; REVERSION PROHIBITED: This Plan has been established for the exclusive benefit of Participants and their Beneficiaries. Under no circumstances may any funds contributed to or held by the Trustee at any time revert to or be used for or enjoyed by the Company except to the extent permitted by law.
- 12.6 GENDER AND NUMBER: The masculine gender will be deemed to include the feminine gender, and the singular will be deemed to include the plural.

Notwithstanding any other provision of the Plan to the contrary, in the event the Plan is deemed to be a "Top Heavy Plan" (as defined in Section 13.1[i]) for any Plan Year, the provisions contained in Section 13.4 with respect to Company Contributions shall be applicable with respect to such Plan Year

- 13.1 DEFINITIONS: For purposes of this Article XIII, the following definitions shall apply:
 - [a] "Determination Date" means, with respect to any Plan Year, the last day of the preceding Plan Year, except that for the first Plan Year the "Determination Date" shall be the last day of such first Plan Year.
 - [b] "Key Employee" means an Employee or former Employee who, at any time during the current Plan Year or any of the four preceding Plan Years, is a key employee pursuant to the provisions of Section 416(i)(1) of the Code, and any Beneficiary of such an Employee or former Employee.
 - [c] "Non-Key Employee" means an Employee or former Employee who is not a Key Employee, and his Beneficiary in the event of his death.
 - [d] "Permissive Aggregation Group" means the group of tax-qualified plans of the Company or a Related Corporation consisting of:
 - [1] the plans in the Required Aggregation Group; plus
 - [2] one or more plans designated from time to time by the Plan Administrator that are not part of the Required Aggregation Group but that satisfy the requirements of Sections 401(a)(4) and 410 of the Code when considered with the Required Aggregation Group.

-36-

- [e] "Required Aggregation Group" means the group of tax-qualified plans of the Company or a Related Corporation consisting of:
 - [1] each plan in which a Key Employee participates; and
 - [2] each other plan which enables a plan in which a Key Employee participates to meet the requirements of Sections 401(a)(d) and 410 of the Code, including any such plan that has been terminated if it was maintained during the five-year period ending on the Determination Date and would, but for the fact that it terminated, be part of a Required Aggregation Group.
- [f] "Super Top Heavy Group" means, with respect to a particular Plan Year, a Required or Permissive Aggregation Group that, as of the Determination Date, would qualify as a Top Heavy Group under the definition in subsection [h] below with "90 percent" substituted for "60 percent" each place where "60 percent" appears in such definition.
- [g] "Super Top Heavy Plan" means, with respect to a particular Plan Year, a plan that, as of the Determination Date, would qualify as a Top Heavy Plan under the definition in subsection [i] below with "90 percent" substituted for "60 percent" each place where "60 percent" appears in such definition. A plan shall also be a "Super Top Heavy Plan" if it is part of a Super Top Heavy Group.
- [h] "Top Heavy Group" means, with respect to a particular Plan Year, a Required or Permissive Aggregation Group if the sum, as of the Determination Date, of the present value of the cumulative accrued benefits for Key Employees under all defined benefit plans included in such Group and the aggregate of the account balances of Key Employees under all defined contribution plans included in such Group exceeds 60 percent of a similar sum determined for all employees covered by the plans included in such Group.
- [i] "Top Heavy Plan" means, with respect to a particular Plan Year,(i) in the case of a defined contribution plan (including any

simplified employee pension plan), a plan for which, as of the Determination Date, the aggregate of the accounts (within the meaning of Section 416(g) of the Code and the regulations and rulings promulgated thereunder) of Key Employees exceeds 60 percent of the aggregate of the accounts of all participants under the plan, with accounts valued as of the relevant "Valuation Date" (as defined in subsection [j] below), (ii) in the case of a defined benefit plan, a plan for which, as of the Determination Date, the present value of the cumulative accrued benefits payable under the plan (within the meaning of Section 416(g) of the Code and regulations and rulings promulgated thereunder) to Key Employees exceeds 60 percent of the present value of the cumulative accrued benefits under the plan for all employees, with the present value of cumulative accrued benefits to be determined under the accrual method uniformly used under all plans maintained by the Company or a Related Corporation or, if no such method exists, under the slowest accrual method permitted under the fractional accrual rate of Section 411(b)(1)(C) of the Code, and (iii) any plan (including any simplified employee pension plan)

-37-

included in a Required Aggregation Group that is a Top Heavy Group. For purposes of this subsection [i], the accounts and accrued benefits of any employee who has not performed services for the Company or a Related Corporation during the five-year period ending on the Determination Date shall be disregarded. Furthermore, the present value of cumulative accrued benefits under a defined benefit plan for purposes of top heavy determinations shall be calculated using the actuarial assumptions otherwise employed under such plan, except that the same actuarial assumptions shall be used for all plans within a Required or Permissive Aggregation Group. A Participant's interest in the Plan attributable to any Rollover Contributions, except Rollover Contributions made from a plan maintained by the Company or a Related Corporation, shall not be considered in determining whether a plan is top heavy. Notwithstanding the foregoing, if a plan is included in a Required or Permissive Aggregation Group that is not a Top Heavy Group, such plan shall not be a Top Heavy Plan.

- [j] "Valuation Date" means, with respect to a Determination Date and for purposes of this Article XIII only, (i) in the case of a defined benefit plan, the most recent date for computing plan costs for minimum funding purposes, and (ii) in the case of a defined contribution plan, the most recent date on which plan assets are valued for purposes of determining the value of account balances.
- 13.2 COMPENSATION AND LIMITATION THEREON: For purposes of this Article XIII, "compensation" shall mean "compensation" as defined in Section 4.5; provided, however, that for any Plan Year that the Plan is a Top Heavy Plan, the compensation taken into account under the Plan for any Participant shall not exceed \$150,000 (subject to adjustment for cost-of-living annually by the Secretary of the Treasury as provided in Sections 401(a) (17) (B) and 416(d) (2) of the Code).
- 13.3 VESTING REQUIREMENTS: In any Plan Year that the Plan is a Top Heavy Plan, each Participant's entire interest in his Account shall be fully vested and nonforfeitable. In the event the Plan ceases to be a Top Heavy Plan for any Plan Year subsequent to a Plan Year in which the Plan was a Top Heavy Plan, the Participant's interest in his Account that has become fully vested in accordance with the preceding sentence shall remain fully vested.
- MINIMUM ALLOCATION: Each Participant who, on the last day of a Plan Year in which the Plan is a Top Heavy Plan, (i) is a Non-Key Employee and (ii) does not participate in a defined benefit plan that is part of a Required Aggregation Group, shall receive a minimum allocation of Company Matching Contributions for such Plan Year equal to a certain percentage (as hereinafter set forth) of his compensation received during such Plan Year. Such percentage shall be equal to the lesser of three percent or the highest percentage at which Company Matching Contributions are allocated to the Account of any Key Employee for such Plan Year (when expressed as a percentage of such Key Employee's compensation for the Plan Year). To the extent necessary to provide this minimum allocation, the allocation of Company Matching Contributions to Accounts of Key Employees shall be reduced proportionately. Notwithstanding the foregoing, an individual who, on the last day of any Plan Year during which the Plan is a Top Heavy Plan, is a participant in both a defined benefit plan and a defined contribution plan maintained by the Company or a Related Corporation shall receive the minimum

allocation referred to above, except that "seven and one-half percent" shall be substituted for "three percent"; provided, however, that (i) if such individual receives a minimum benefit with respect to such Plan Year under a defined benefit plan maintained by the Company or a Related Corporation, such individual shall not receive a minimum benefit under this Plan; and (ii) if the Plan is a Super Top Heavy Plan, the limitation in Section 415(e) of the Code shall be applied by substituting "1.0" for "1.25", and the minimum allocation referred to in this Section 13.4 shall be applied by substituting "five percent" for "seven and one-half percent".

ARTICLE XIV

SPECIAL PROVISIONS REGARDING ACQUISITION EMPLOYEES

- 14.1 GENERAL: This Article XIV contains special provisions regarding employees of Northshore Mining Company (formerly Cyprus Northshore Mining Corporation) and employees of Silver Bay Power Company (formerly Cyprus Silver Bay Power Corporation) on the day preceding the Effective Date (hereinafter, "Acquisition Employees") who were also participants in the Cyprus Amax Minerals Company Savings Plan & Trust (the "Cyprus Savings Plan") on the day preceding the Effective Date.
- 14.2 EMPLOYMENT: For all purposes of the Plan, the acquisition of Northshore Mining Company and Silver Bay Power Company pursuant to the Stock Purchase Agreement by and between Cleveland-Cliffs Inc and Cliffs Minnesota Minerals Company and Cyprus Amax Minerals Company, as amended the ("Stock Purchase Agreement") shall not be deemed a Termination of Employment, an employment commencement date, a reemployment commencement date or otherwise a break in employment or service with respect to Acquisition Employees.
- TRANSFER OF ASSETS AND LIABILITIES: Effective as of the Effective Date (the "Transfer Date"), all liabilities of the Cyprus Savings Plan with respect to Acquisition Employees shall be transferred to the Plan from the Cyprus Savings Plan; provided, however, that on or as soon as practicable after the Transfer Date, assets of the Cyprus Savings Plan attributable to such Acquisition Employees' liabilities so transferred shall be transferred from the Cyprus Savings Plan to the Plan in an amount equal to the accounts of such Acquisition Employees under the Cyprus Savings Plan on the "Closing Date" (as "Closing Date" is defined in the Stock Purchase Agreement), which accounts shall be adjusted for gains and losses for the period commencing on the day following the Closing Date and ending on the day assets are actually transferred from the Cyprus Savings Plan to the Plan. Anything contained herein to the contrary notwithstanding, liabilities of the Cyprus Savings Plan with respect to Acquisition Employees that are not transferred to the Plan on the Transfer Date (if any) shall be transferred to the Plan as soon thereafter as administratively possible (the "Second Transfer Date"). The amount of assets of the Cyprus Savings Plan attributable to Acquisition Employees' liabilities so transferred on the Second Transfer Date shall be determined and transferred as hereinbefore set forth in this Section 14.3. To the extent possible, assets transferred from the Cyprus Savings Plan to the Plan shall be transferred in kind. Any applications, elections and waivers under the Cyprus Savings Plan applicable to assets transferred from the Cyprus Savings Plan to the Plan shall continue to be applicable hereunder, unless the Acquisition Employee revises such application,

-39-

election or waiver. Assets transferred to the Plan from the Cyprus Savings Plan pursuant to the provisions of this Section 14.3 shall become assets of the Plan to be held by the Trustee in the Trust Fund.

14.4 CONTINUATION OF PORTION OF CYPRUS SAVINGS PLAN: The Plan shall be deemed a continuation of the portion of the Cyprus Savings Plan transferred to the Plan with respect to Acquisition Employees; provided, however, that the Plan shall only be deemed a continuation of the aspects of the Cyprus Savings Plan governed by Sections 401(a), 401(k) and 401(m) of the Code.

* * *

NORTHSHORE MINING COMPANY

By: /s/ Cynthia B. Bezik

Title Treasurer

And: /s/ J. E. Lenhard

Title Secretary

SILVER BAY POWER COMPANY

By: /s/ Cynthia B. Bezik

Title Treasurer

And: /s/ J. E. Lenhard

Title Secretary

HR456

- 40 -EXHIBIT "A"

NORTHSHORE MINING COMPANY AND SILVER BAY POWER COMPANY RETIREMENT SAVINGS PLAN

The following companies are participating Companies under the Plan:

Northshore Mining Company Silver Bay Power Company

T. ROWE PRICE TRUST COMPANY

TRUST AGREEMENT

This TRUST AGREEMENT is made this 10 day of OCTOBER, 1994, by and between NORTHSHORE MINING COMPANY hereinafter referred to as the "EMPLOYER,", SILVER BAY POWER COMPANY (a "PARTICIPATING EMPLOYER"), and T. ROWE PRICE TRUST COMPANY, a Maryland limited trust company, hereinafter referred to as the "TRUSTEE."

WITNESSETH

WHEREAS, the Employer has adopted the NORTHSHORE MINING COMPANY AND SILVER BAY POWER COMPANY RETIREMENT SAVINGS PLAN, a defined contribution plan intended to be a qualified plan under Section 401(a) of the Internal Revenue Code ("CODE"), which plan is hereinafter referred to as the "PLAN," and Silver Bay Power Company is a Participating Employer thereunder, for the benefit of all those individuals eligible to participate under the Plan terms (including beneficiaries and alternate payees), hereinafter referred to individually as "PARTICIPANT" and collectively as "PARTICIPANTS;" and

WHEREAS, the Plan provides that the assets thereof be held, in trust, by a trustee, subject to the provisions of a trust agreement to be entered into between the Employer and a trustee or trustees;

NOW THEREFORE, the Employer and the Trustee agree as follows:

ARTICLE I - TRUST FUND

- 1.1 TRUST. The Employer hereby establishes with the Trustee, a trust account or accounts ("ACCOUNTS") consisting of such sums of U.S. currency and such other property acceptable to the Trustee as shall from time to time be contributed to, paid or delivered to the Trustee pursuant to this Trust Agreement at the address specified by the Trustee. All such money and property, all investments and reinvestments made therewith and proceeds thereof, less any payments or distributions made by the Trustee pursuant to the terms of this Trust Agreement are referred to herein as the "TRUST." The Trust shall be held by the Trustee in accordance with the express provisions of this instrument and the requirements of law.
- 1.2 CUSTODY OF TRUST ASSETS. The Trustee is authorized to: (a) hold property hereunder in bearer form or in its own name or the name of its nominee; (b) combine ${}^{\circ}$

Sequential Page 57

certificates representing investments of the Trust with certificates of the same issue held by the Trustee or other fiduciaries; (c) hold securities in definitive form on a segregated or nonsegregated basis or with a correspondent bank or depository (or nominee of such bank or depository); and (d) hold obligations of the United States Government and agencies thereof on a book entry basis at the appropriate Federal Reserve bank, but the books and records of the Trustee shall, at all times, show that all such property and securities are held in trust. The Trustee shall not hold any property or securities hereunder in the same account as any individual property of the Trustee. The Trustee is also authorized to appoint a subcustodian to perform any of the above functions.

1.3 LIMITATIONS OF TRUSTEE'S DUTIES. With respect to its duties hereunder, the Trustee is a non-discretionary trustee and shall have no duty to: (a) determine or enforce payment of any contribution due under the Plan; (b) inquire into the accuracy of any contribution; (c) determine the adequacy of the funding policy adopted by the Employer to meet its obligations under the Plan; (d) look into the propriety of any distribution made under the Plan; or (e) ensure the qualification of the Plan under the Code. The Trustee shall not be deemed to be the administrator, the Plan sponsor or a "named fiduciary" of the Plan as defined in Sections 3(16)(A), 3(16)(B) and 402(a)(2), respectively of the Employee Retirement Income Security Act of 1974 ("ERISA").

ARTICLE II - ACCOUNTS

2.1 ESTABLISHING ACCOUNTS. The Trustee shall open and maintain a trust Account for the Plan. Upon receipt of written instructions from the Employer, the Trustee also shall open and maintain such Participant Accounts and subaccounts as the Employer may direct. The Trustee shall also open and maintain such other subaccounts as may be appropriate or desirable to aid in the administration of the Plan. The Employer shall give written instructions to the Trustee specifying the Participants' Accounts and subaccounts to which contributions and forfeitures (if any) are to be credited, and the amounts of such contributions and forfeitures (if any) which are to be credited to such

2.2 CHARGES AGAINST ACCOUNTS. Upon receipt of written instructions from the Employer, the Trustee shall charge the appropriate Account or subaccount of a Participant for any withdrawals or distributions made under the Plan, for any forfeiture (if any) which may be required under the Plan of unvested interests attributable to Employer contributions and for any fees which may be charged against the Trust assets.

- 2 - ARTICLE III - INVESTMENT OF TRUST ASSETS

- 3.1 INVESTMENT OF TRUST ASSETS. The Trustee shall not have any discretion, and is specifically prohibited from having or exercising any discretion, with respect to the investment of Trust assets. Except as provided in Section 3.3 (Participant Directed Investments) hereof, the Employer shall be the "named fiduciary" and shall be solely responsible for giving the Trustee directions as to the investment and disposition of the Trust assets, including but not limited to quaranteed investment contracts, bank investment contracts, synthetic investment contracts, certificates of deposit and insurance company annuity contracts. The Trustee, unless it has knowledge that an investment direction constitutes a violation of ERISA or any other applicable law, shall be entitled to rely on such direction, and the Trustee shall not review any securities or other assets or make suggestions with respect to the investment, reinvestment, retention or disposition of any Trust assets. The Trustee shall invest and reinvest the Trust's assets only as directed and free from any limitations imposed by state law on investments of trust funds and without distinction between income and principal in any property, including, but not limited to, common and preferred stocks, governmental obligations, equipment, trust certificates, participation certificates, investment companies or trusts (including any investment company or trust which has an investment advisory or other agreement with an affiliate of the Trustee), collateral trust notes, savings and time deposits, commercial paper (including participation in variable amount notes), leasebacks, mortgages and other interests in realty, corporate bonds, debentures, notes and other evidences of indebtedness, secured or unsecured, non-income producing securities or property, options and participation in any group or common trust funds, including any such funds held or maintained by an affiliate of the Trustee, for commingling assets of participating trusts and exempt from Federal income tax withholding, but not limited to, any group or common trust fund which is qualified under the provisions of Section 401(a) of the Code or any successor provisions thereto (the instrument of trust creating any such qualified group or common trust fund, to the extent of the Trust's equitable share thereof, being adopted hereby).
- 3.2 WRITTEN INSTRUCTION. Any action of the Employer pursuant to any provisions of this Agreement shall be in writing from the Employer and the Trustee shall be fully protected in relying upon such written notification as actions of the Employer. The term "EMPLOYER," as used throughout this Agreement, includes any duly authorized designee of the Employer, such as a Plan Administrator, or any individual having apparent authority as such. If written instructions are not received by the Trustee, or if such instructions are received but are deemed by the Trustee to be unclear, upon notice to the Employer, the Trustee may elect to hold all or part of any such contribution in cash, without liability for rising security prices or distributions made, pending receipt by it from the Employer of written instructions or other clarification. If any contributions received by the Trustee from the Employer are less than any minimum which a directed investment requires, the Trustee may hold the specified

- 3 -

portion of such contributions in cash, without interest, until such time as the proper amount has been contributed so that the directed investment may be made. The Trustee shall receive all directions or instructions in writing provided that the Trustee may accept oral directions for purchases or sales from the Employer or Participant with subsequent written confirmation.

3.3 PARTICIPANT DIRECTED INVESTMENTS. When so instructed by the Employer, the Trustee shall invest all or any portion of the individual Accounts of any Participant as directed by said Participant. Such directed investments shall be accounted for separately for each Participant. The Employer or its designee shall have the duty to select and monitor all investment options made available to Participants under the Plan. The Employer shall ensure that all Participants who are entitled to direct the investment of assets in their Accounts previously received or receive a copy of all material describing such investment options that is required by law. Delivery of investment directions by the Employer in accordance with the instructions of a Participant or by the Participant directly to the Trustee shall entitle the Trustee to assume that the Participant has received all such descriptive material. Each Participant who directs the investment of his Accounts shall be solely and absolutely

responsible for the investment, or reinvestment of any such directed Plan investment held on his behalf in the Trust, and, except as otherwise provided herein, the Trustee shall not question any such direction, review any securities or other such assets, or make suggestions with respect to the investment, reinvestment, retention or disposition of any such assets. The Trustee shall not have any liability or responsibility for diversification of such assets, for any loss to or depreciation of such assets because of the purchase, retention or sale of assets in accordance with a Participant's direction, and the Participant shall have sole responsibility for the overall diversification, liquidity and prudence of the investments of his Accounts. If a Participant fails to direct the investments of his Accounts, the Trustee shall invest his Accounts in accordance with the written directions of the Employer.

3.4 EMPLOYER DIRECTED INVESTMENTS. The Employer, by written direction to the Trustee, is authorized to designate all or a portion of the Trust assets of which the Employer will direct investments, and the Trustee may segregate such assets into one or more separate accounts or administer the Trust as one account. In the event the Employer shall employ or appoint an investment advisor to direct the Trustee with respect to a portion of the Trust, the Employer will notify the Trustee in writing of the appointment of an investment advisor, including its name and address. Whether or not the Trust is segregated into separate accounts, the Trustee shall invest such portion of the Trust as directed by the Employer or its duly appointed investment advisor only to the extent that such instruction is consistent with ERISA and any other applicable legal authority. The Trustee shall have no duty to question any action or direction of the Employer or investment advisor or any failure of the Employer or investment advisor to give directions, or to review the securities or other investments which are held pursuant to the Employer's or investment advisor's directions, or to

- 4 -

make suggestions to the Employer or investment advisor as to the investment, reinvestment, retention or disposition of any such assets. The Trustee shall not have any liability or responsibility for diversification of such assets, or for any loss to or the depreciation of such assets because of the purchase, retention or sale of assets in accordance with the Employer's or investment advisors direction. The Employer shall have responsibility for the overall diversification of the Trust.

- 3.5 TRUSTEE'S LIABILITY WITH RESPECT TO EMPLOYER OR PARTICIPANT DIRECTED ACCOUNTS. The Trustee shall not be liable for, and the Employer will indemnify and hold harmless the Trustee (including its affiliates, representatives and agents) from and against, any liability or expense (including counsel fees) because of: (a) any investment action taken or omitted by the Trustee in accordance with any direction of the Employer or a Participant; or (b) any investment inaction in the absence of directions from the Employer or a Participant; or (c) any investment action taken by the Trustee pursuant to an order to purchase or sell securities placed by the Employer or a Participant directly with a broker, dealer or issuer.
- 3.6 LIMITATIONS ON INVESTMENTS. Notwithstanding any other provision of this Trust Agreement to the contrary:
- (a) The Trustee may establish such reasonable rules and regulations, applied on a uniform basis to all Participants, with respect to the requirements for, and the form and manner of, effecting any transaction with respect to Participant directed investments as the Trustee shall determine to be consistent with the purposes of the Plan. Any such rules and regulations shall be binding upon all persons interested in the Trust.
- (b) In no event shall the Trustee engage in any transactions that would be prohibited under ERISA.
- 3.7 "KNOWLEDGE" OF TRUSTEE. It is understood that although, when the Trustee is subject to the direction of the Employer or a Participant, the Trustee will perform certain ministerial duties ("MINISTERIAL DUTIES") with respect to the portion of the Trust subject to such direction, such duties do not involve the exercise of any discretionary authority to manage or control Trust assets. Such Ministerial Duties will be performed in the normal course of business by employees of the Trustee, its affiliates, or agents who may be unfamiliar with investment management. It is agreed that the Trustee is not undertaking any duty or obligation, express or implied, to review, and will not be deemed to have any knowledge of or responsibility with respect to, any transaction involving the investment of the Trust as a result of the performance of these Ministerial Duties. Therefore, in the event that "knowledge" of the Trustee shall be a prerequisite to imposing a duty upon or determining liability of the Trustee under the Plan, this Trust Agreement or any law regulating the conduct of directed trustees with respect to the investment of trust assets, as a result of any act or omission of the

Employer, or any Participant, or as the result of any transaction engaged in by any of them, then the receipt and processing of investment orders and other documents relating to Trust assets by an employee of the Trustee or its affiliates or agents engaged in the performance of purely Ministerial Duties shall not constitute "knowledge" of the Trustee.

ARTICLE IV - DUTIES OF THE TRUSTEE

- $4.1\,$ DUTIES OF THE TRUSTEE. The Trustee is authorized and empowered with respect to the Trust:
- (a) To make, execute, acknowledge and deliver any and all documents of transfer and conveyance and any and all other instruments that may be necessary or appropriate to carry out the powers herein granted.
- (b) To register any investment held in the Trust in the name of the Trustee or in the name of a nominee, and to hold any investment in bearer form, but the books and records of the Trustee shall at all times show that all such investments are part of the Trust.
- (c) To employ suitable agents and counsel (who may also be agents and/or counsel for the Employer) and to pay their reasonable expenses and compensation.
- (d) To borrow or raise monies for the purpose of the Trust from any source and, for any sum borrowed, to issue its promissory note as Trustee and to secure the repayment thereof by pledging all or any part of the Trust, but nothing contained herein shall obligate the Trustee to render itself liable individually for the amount of any such borrowing; and no person loaning money to the Trustee shall be bound to see the application of money loaned or to inquire into the validity or propriety of any such borrowing.

Each and all of the foregoing powers may be exercised without a court order or approval. No one dealing with the Trustee need inquire concerning the validity or propriety of anything that is done or need see the application of any money paid or property transferred to or upon the order of the Trustee.

4.2 GENERAL POWERS. The Trustee shall have all of the powers necessary or desirable to do all acts, take such procedures and exercise all such rights and privileges, whether or not expressly authorized herein, which it may deem necessary or proper for the protection of the Trust and to accomplish any action provided for in this Trust Agreement.

- 6 -

- 4.3 VALUATION OF TRUST. The Trustee, as of the valuation date, and at such other time or times as is necessary, shall determine the net worth of the assets of the Trust. The Trustee may adopt such methods of valuation as it deems advisable.
- 4.4 TRUST RECORDS. The Trustee shall keep accurate and detailed records of all receipts, investments, disbursements and other transactions required to be performed hereunder with respect to the Trust. The Trustee agrees to treat as confidential all records and other information relative to the Trust and Participant Accounts. The Trustee shall not disclose such records and other information to third parties except to the extent required by law or as requested in writing by the Employer.
- 4.5 DISTRIBUTIONS. At the direction of the Employer, the Trustee shall make distributions from the Trust to the Employer for the benefit of the Participants and, to the extent agreed to by the Trustee, shall make distributions directly to the Participants. The Trustee shall not be liable or responsible for any errors made by the Employer with respect to distributions. The Trustee shall be entitled to rely conclusively upon the Employer's directions. Notwithstanding any other provision of the Trust Agreement, the Trustee may condition its delivery, transfer or distribution of any assets upon the Trustee's receiving satisfactory assurances that the approval of appropriate governmental agencies or other authorities have been secured and that all notice and other procedures required by applicable law have been satisfied.
- 4.6 TRUSTEE'S FEES. The Trustee's fees for performing its duties hereunder shall be such reasonable amounts as shall be established by it from time to time. The Trustee shall furnish to the Employer its current schedule of fees and give written notice to the Employer whenever its fees are changed or revised. Such fees, any taxes of any kind whatsoever which may be levied or

assessed upon the Trust, and any expenses incurred by the Trustee in the performance of its duties, including fees for legal services rendered to the Trustee, shall, unless paid by the Employer, be paid from the Trust.

- 4.7 DUTIES NOT ASSIGNED. The duties of the Trustee with respect to the Trust are limited to those assumed by the Trustee under the terms of this Trust Agreement. The Trustee shall not be responsible for filing reports, returns or disclosures with any government agency except as may otherwise be required by its duties as Trustee under applicable law.
- 4.8 STANDARDS FOR THE TRUSTEE'S POWERS. Notwithstanding any other provision of this Trust Agreement, the Trustee shall discharge its duties hereunder solely in the interest of the Participants and for the exclusive purpose of providing benefits to the Participants and defraying reasonable expenses of administering the Trust, with skill, care, prudence and diligence under the circumstances then prevailing that a prudent man acting in a like capacity and familiar with such matters would use in the conduct of an enterprise of a like character and with like aims. The Trustee shall perform its

- 7 -

duties in accordance with this Trust Agreement insofar as this Trust Agreement is consistent with the provisions of ERISA. To the extent not prohibited by ERISA, the Trustee shall not be responsible in any way for any action or omission of the Employer with respect to the performance of the Employer's duties and obligations set forth in this Agreement and in the Plan. The Trustee may rely upon such information, direction, action or inaction of the Employer as being proper under the Plan or the Trust Agreement and is not required to inquire into the propriety of any such information, direction, action or inaction. To the extent not prohibited by ERISA, the Trustee shall not be responsible for any action or omission of any of its agents, or with respect to reliance upon advice of its counsel (whether or not such counsel is also counsel to the Employer), provided that such agents or counsel were prudently chosen by the Trustee and that the Trustee relied in good faith upon the action of such agent or the advice of such counsel.

ARTICLE V - DUTIES OF THE EMPLOYER

- 5.1 DUTIES OF THE EMPLOYER. It is understood that the Employer shall be responsible for the performance of the following functions with respect to the $T_{\rm rust}$.
- (a) Transmitting all Trust contributions made by or on behalf of each Participant in accordance with the instructions of each Participant to the Trustee at such times and in such manner as is mutually agreed between the Employer and the Trustee.
- (b) Providing to the Trustee, on a timely basis, all Participant enrollment forms and such other forms relating to Participant Accounts, including any subsequent amendments.
- (c) Determining that the contributions made by or on the behalf of each Participant are in accordance with any applicable Federal and state law and regulations.
- (d) Assuring that the Plan maintains qualified status under applicable provisions of the Code.
- 5.2 BONDING. The Employer agrees to obtain and maintain a fiduciary bond and to include as those covered by such bond the employees of the Employer, the Plan Administrator and the Trustee, including any of its employees, officers and agents required by law to be so covered. The cost of any such bond shall be paid by the Employer.
- 5.3 INFORMATION AND DATA TO BE FURNISHED TO THE TRUSTEE. The Employer shall furnish the Trustee with such information and data relevant to the Plan as is

- 8 -

necessary for the Trustee to properly perform its duties assumed hereunder, including but not limited to a copy of the Plan's qualification letter from the Internal Revenue Service.

5.4 LIMITATION OF DUTIES. Neither the Employer nor any of its officers, directors, partners or agents shall have any duties or obligations with respect to this Trust Agreement, except those expressly set forth herein, in the Plan and in ERISA.

 $5.5\,$ QUALIFIED DOMESTIC RELATIONS ORDERS. It shall be the responsibility of the Employer to determine whether any domestic relations order is "qualified" in accordance with Code Section 414(p). The Trustee will act only as directed by the Employer with respect to the payment of benefits to an alternate payee under any qualified domestic relations order.

ARTICLE VI - TERMINATION OF TRUST AGREEMENT

6.1 RESIGNATION OR REMOVAL OF TRUSTEE. The Trustee may resign at any time upon thirty days prior written notice to the Employer and may be removed by the Employer at any time upon thirty days' prior written notice to the Trustee. Upon resignation or removal of the Trustee, the Employer shall appoint a successor trustee. Upon receipt by the Trustee of written acceptance of such appointment by the successor trustee, the Trustee shall transfer and pay over to the successor the assets of the Trust and all records (or copies) pertaining thereto. The Trustee is authorized, however, to reserve such sum of money or property as it may deem advisable for payment of all fees, compensation, costs and expenses, or for payment of any liabilities constituting a charge on or against the assets of the Trust or on or against the Trustee, with any balance of such reserve remaining after payment of all such items to be paid over to the successor trustee. To the extent not prohibited by ERISA, upon the assignment, transfer and payment over of the assets of the Trust, and obtaining a receipt thereof from the successor trustee, the Trustee shall be released and discharged from any and all claims, demands, duties and obligations arising out of the Trust and its management thereof, excepting claims based only upon the Trustee's willful misconduct or not acting a manner consistent with that of a prudent professional. The successor trustee shall hold the assets paid over to it under the terms similar to those of this Trust Agreement under a trust that will qualify under Section 401(a) of the Code. If within thirty days after the Trustee's resignation or removal, the Employer has not appointed a successor trustee which has accepted such appointment, the Trustee may bring an appropriate action or proceeding for leave to deposit the assets and cash in a court of competent jurisdiction. The Trustee shall be reimbursed by the Employer for all costs and expenses of the action or proceeding including, without limitation, reasonable attorney's fees and disbursements.

- 9 -

6.2 TERMINATION OF THE TRUST. Subject to the right of the Trustee to terminate the Trust, this Trust shall continue as to the Employer so long as the Plan is in full force and effect. If the Plan ceases to be in full force and effect, this Trust shall thereupon terminate unless expressly extended by the Employer.

ARTICLE VII - MISCELLANEOUS

- 7.1 PURPOSE. This Trust has been established for the exclusive benefit of the Plan's Participants. Except as provided herein, it shall be impossible at any time prior to the satisfaction of all liabilities to the Participants for any part of the principal or income of the Trust, other than such part as is required to pay taxes, administrative expenses or refund contributions as provided herein, to be paid or diverted to the Employer or to be used for any purpose whatsoever other than for the exclusive benefit of the Participants.
- 7.2 INDEMNIFICATION. The Employer shall indemnify and hold harmless the Trustee (including affiliates, employees, representatives and agents) from and against any liability, cost or other expense, including, but not limited to, the payment of attorneys' fees which the Trustee may incur in connection with this Trust Agreement or the Plan unless such liability, cost or expense arises from the Trustee's own willful misconduct or not acting in a manner consistent with that of a prudent professional. The Trustee shall not be obligated or expected to commence or defend any legal action or proceeding in connection with this Agreement unless agreed upon in writing by the Trustee and Employer and unless the Trustee is fully indemnified for doing so to its satisfaction.
- 7.3 CONFLICT WITH THE PLAN DOCUMENT. In the event of any conflict between the provisions of the Plan document, as they pertain to the duties of the Trustee, and this Trust Agreement, the provisions of this Trust Agreement shall prevail.
- 7.4 CONSTRUCTION. Whenever used in this Trust Agreement, unless the context indicates otherwise, the singular shall include the plural, the plural shall include the singular, and the male gender shall include the female gender.
- 7.5 HEADINGS. Headings in this Trust Agreement are inserted solely for convenience or reference and shall neither constitute a part of this Agreement, nor affect its meaning, construction or intent.
- 7.6 SEVERABILITY. If any provision of this Trust Agreement is held invalid or unenforceable, such invalidity or unenforceability shall not affect any other provision, and this Agreement shall be construed and enforced as if such

- 10 -

- 7.7 RETURN OF CONTRIBUTIONS. Contributions are conditioned on initial qualification of the Plan under Section 401(a) of the Code, and if the Plan and Trust do not qualify, the Trustee shall return such contributions to the Employer upon the Employer's written direction. The Trustee shall also return amounts to the Employer upon the Employer's written direction due to a "mistake of fact" as described in Section 403(c) of ERISA. Contributions made by the Employer by "mistake of fact" shall revert and be paid to the Employer within one year after the payment of such mistaken contributions, if the Employer so directs the Trustee in writing. In making such a return of assets to the Employer, the Trustee may accept the Employer's written direction as its warranty that such payment is provided for in the Plan and complies with such plan provision and ERISA Section 403(c), and the Trustee need make no further investigation.
- 7.8 VOTING. Except to the extent provided in subparagraph (i) and (ii) of this section, the Employer shall direct the Trustee as to the manner in which it shall:
- (a) vote in person or by proxy, general or special, any securities held in the Trust ;
 - (b) exercise conversion privileges, subscription rights and other options; and
- (c) participate in or dissent from reorganizations, tender offers or other changes in property rights.
- (i) The Trustee shall exercise all voting or tender offer rights with respect to any qualifying employer securities, as defined in Section 407(d)(5) of ERISA (individually, "QUALIFYING EMPLOYER SECURITY" and collectively, "QUALIFYING EMPLOYER SECURITIES") held by it in accordance with instructions from Participants. Each Participant shall be a named fiduciary within the meaning of Section 403(a)(1) of ERISA for the purpose of directing the voting and tendering of Qualifying Employer Securities allocated to his Account. Each Participant may direct the Trustee, confidentially, how to vote or whether or not to tender the Qualifying Employer Securities representing shares allocated to his Participant Account. Upon timely receipt of direction, the Trustee shall vote or tender all such shares of Qualifying Employer Securities as directed by the Participants. The Employer shall direct the Trustee as to voting of shares of Qualifying Employer Securities for which no Participant direction is received. The Trustee shall use reasonable procedures to inform Participants as to what action will be taken in the absence of such affirmative instructions. In the case of a tender offer or other right or option with respect to Qualifying Employer Securities, a Participant who does not issue valid directions to the Trustee to sell, offer to sell, exchange or otherwise dispose of such Qualifying Employer Securities shall be deemed to have directed the Trustee that such shares allocated to his Participant Account remain invested in Qualifying Employer Securities. The Employer shall provide the

- 11 -

with all information and assistance that the $Trustee\ may$ reasonably request in order for the $Trustee\ to$ perform its duties hereunder.

- (ii) Notwithstanding the foregoing, the Trustee shall follow any directions of the Employer or Participants in the performance of these functions only to the extent that following such directions would not violate the provisions of ERISA.
- 7.9 NONALIENATION OF BENEFITS. No rights or claims to any of the monies or other assets of the Trust shall be assignable, nor shall such rights or claims be subject to garnishment, attachment, execution or levy of any kind; and any attempt to transfer, assign or pledge the same, except as specifically permitted by law, shall not be recognized by the Trustee.
- 7.10 AMENDMENTS. The Employer and the Trustee may amend this Agreement at any time by a written agreement between them; provided, however, that no such amendment shall make it possible for any part of the corpus or income of the Fund to be used or diverted to purposes other than the exclusive benefit of Participants and defraying reasonable expenses of administering the Plan and Trust.
- 7.11 INSPECTION OF PLAN RECORDS BY EMPLOYER. The Trustee agrees to permit the Employer to inspect the records of the Trust maintained by the Trustee during regular business hours and to permit the Employer to audit the same upon the

giving of reasonable notice to the Trustee. The Trustee further agrees that it will provide the Employer with information and records that the Employer may reasonably require in order to perform audits of said records.

- 7.12 LAW GOVERNING. This Agreement shall be administered, construed and enforced according to the laws of the State of Maryland and applicable Federal law.
- 7.13 MERGER, CONSOLIDATION OR TRANSFER. In the event of the merger, consolidation or transfer of any portion of the Trust to a trust fund held under any other plan, the Trustee shall dispose of all or part, as the case may be, of the Trust, in accordance with the written directions of the Employer, subject to the right of the Trustee to reserve funds as provided in Section 6.1 hereof.
- 7.14 TRUSTEE AS SUCCESSOR TRUSTEE. If the Trustee is acting as a successor trustee with respect to the Trust, the Employer shall indemnify the Trustee against all liabilities with respect to the Trust arising prior to the appointment of the Trustee and its acceptance thereof.
- 7.15 SUCCESSORS AND ASSIGNS. This Agreement shall be binding upon the successor and assigns of the parties hereto.

- 12 -

7.16 EFFECTIVE DATE. This Agreement shall be effective as of the date of transfer to T. Rowe Price Trust Company of the assets which are to be held in trust pursuant to this Agreement but in any event no earlier than October 1, 1994.

IN WITNESS WHEREOF, the Employer and the Trustee have caused their duly authorized officers to execute this Agreement on the date hereinabove written.

Secretary Title

SILVER BAY POWER COMPANY

NORTHSHORE MINING COMPANY

ATTEST:

By: /s/ Cynthia B. Bezik

T. ROWE PRICE TRUST COMPANY

ATTEST:

By: /s/ Regina Pizzonia
/s/ Anne E. Carbaugh
----Vice President
----Title

Consent of Independent Auditors

We consent to the incorporation by reference in the Registration Statement (Form S-8) and related Prospectus of Cleveland-Cliffs Inc pertaining to the Northshore Mining Company and Silver Bay Power Company Retirement Savings Plan of our reports (a) dated February 14, 1994, with respect to the consolidated financial statements and schedules of Cleveland-Cliffs Inc and consolidated subsidiaries included in its Annual Report (Form 10-K) and (b) dated February 14, 1994, with respect to the financial statements and schedules of Tilden Mining Company included in the Annual Report (Form 10-K) of Cleveland-Cliffs Inc, both for the year ended December 31, 1993, filed with the Securities and Exchange Commission.

ERNST & YOUNG LLP

Cleveland, Ohio November 29, 1994

Sequential Page 70

DIRECTORS AND OFFICERS OF CLEVELAND-CLIFFS INC

POWER OF ATTORNEY

REGISTRATION STATEMENT ON FORM S-8

KNOW ALL MEN BY THESE PRESENTS, that each of the undersigned directors and officers of Cleveland-Cliffs Inc, an Ohio corporation (the "Company"), hereby (1) constitutes and appoints John S. Brinzo, Frank L. Hartman and John E. Lenhard, collectively and individually, as his or her agent and attorney-in-fact with full power of substitution and resubstitution to (a) sign and file on his or her behalf and in his or her name, place and stead in any and all capacities (i) a Registration Statement on Form S-8 (the "Registration Statement") with respect to the registration under the Securities Act of 1933, as amended, of participation interests issuable under the Northshore Mining Company and Silver Bay Power Company Retirement Savings Plan (the "Plan") and up to 200,000 shares of the Company's common stock, par value \$1.00 per share, for issuance under the Plan, (ii) any and all amendments, including post-effective amendments, and exhibits to the Registration Statement and (iii) any and all applications or other documents to be filed with the Securities and Exchange Commission or any state securities commission or other regulatory authority with respect to the securities covered by the Registration Statement and (b) do and perform any and all other acts and deeds whatsoever that may be necessary or required in the premises and (2) ratifies and approves any and all actions that may be taken pursuant hereto by any of the above-named agents and attorneys-in-fact or their substitutes.

Executed as of the 28th day of November, 1994.

<TABLE>

/s/M. T. Moore

M. T. Moore Chairman, Presid

Chairman, President and Chief Executive Officer and Director (Principal Executive Officer)

/s/R. S. Colman

R. S. Colman, Director

/s/J. D. Ireland, III

J. D. Ireland, III, Director

/s/G. F. Joklik

G. F. Joklik, Director

/s/E. B. Jones

- -----

E. B. Jones, Director

/s/Leslie L. Kanuk

L. L. Kanuk, Director

/s/Stephen B. Oresman

S. B. Oresman, Director

</TABLE>

<C>

/s/A. Schwartz

A. Schwartz, Director

/s/S. K. Scovil

S. K. Scovil, Director

/s/J. H. Wade

.

J. H. Wade, Director

/s/A. W. Whitehouse

A. W. Whitehouse, Director

/s/J. S. Brinzo

J. S. Brinzo

Senior Executive - Finance (Principal Financial Officer)

/s/R. Emmet

R. Emmet

Vice President and Controller (Principal Accounting Officer)

Sequential Page 71